

# Uncertainty and Bias in Global to Regional Scale Assessments of Current and Future Coastal Flood Risk

J. Hinkel, L. Feyen, M. Hemer, Gonéri Le Cozannet, D. Lincke, M. Marcos, L.

Mentaschi, J L Merkens, H. de Moel, S. Muis, et al.

# ▶ To cite this version:

J. Hinkel, L. Feyen, M. Hemer, Gonéri Le Cozannet, D. Lincke, et al.. Uncertainty and Bias in Global to Regional Scale Assessments of Current and Future Coastal Flood Risk. Earth's Future, 2021, 9 (7), 10.1029/2020ef001882 . hal-03745681

# HAL Id: hal-03745681 https://brgm.hal.science/hal-03745681

Submitted on 4 Aug 2022

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



# **Earth's Future**

# **RESEARCH ARTICLE**

10.1029/2020EF001882

#### **Key Points:**

- We present the first comparison of uncertainties in global to worldregional scale assessments of current and future coastal flood risk
- The largest uncertainty relates to future coastal adaptation, which can influence future coastal flood risk by factors of 20–27
- Uncertainties in socioeconomic development, elevation data, defense levels, emissions and ice sheets can affect risks by factors of 2–6

#### Correspondence to:

J. Hinkel, hinkel@globalclimateforum.org

#### **Citation:**

Hinkel, J., Feyen, L., Hemer, M., Le Cozannet, G., Lincke, D., Marcos, M., et al. (2021). Uncertainty and bias in global to regional scale assessments of current and future coastal flood risk. *Earth's Future*, 9, e2020EF001882. https://doi.org/10.1029/2020EF001882

Received 29 OCT 2020 Accepted 1 JUN 2021

#### **Author Contributions:**

Conceptualization: J. Hinkel Funding acquisition: J. Hinkel Methodology: J. Hinkel Project Administration: J. Hinkel Validation: J. Hinkel Visualization: J. Hinkel Writing – original draft: J. Hinkel, L. Feyen, M. Hemer, G. Cozannet, D. Lincke, M. Marcos, J. L. Merkens, H. de Moel, S. Muis, R. J. Nicholls, A. T. Vafeidis, M. I. Vousdoukas, T. Wahl, P. J. Ward, C. Wolff

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

# Uncertainty and Bias in Global to Regional Scale Assessments of Current and Future Coastal Flood Risk

J. Hinkel<sup>1,2</sup>, L. Feyen<sup>3</sup>, M. Hemer<sup>4</sup>, G. Le Cozannet<sup>5</sup>, D. Lincke<sup>1</sup>, M. Marcos<sup>6,7</sup>, L. Mentaschi<sup>3,8</sup>, J. L. Merkens<sup>9</sup>, H. de Moel<sup>10</sup>, S. Muis<sup>10,11</sup>, R. J. Nicholls<sup>12</sup>, A. T. Vafeidis<sup>9</sup>, R. S. W. van de Wal<sup>13</sup>, M. I. Vousdoukas<sup>3</sup>, T. Wahl<sup>14</sup>, P. J. Ward<sup>10</sup>, and C. Wolff<sup>9</sup>

<sup>1</sup>Global Climate Forum (GCF), Berlin, Germany, <sup>2</sup>Division of Resource Economics, Albrecht Daniel Thaer-Institute and Berlin Workshop in Institutional Analysis of Social-Ecological Systems (WINS), Humboldt-University, Berlin, Germany, <sup>3</sup>European Commission, Joint Research Centre (JRC), Ispra, Italy, <sup>4</sup>CSIRO Oceans and Atmosphere, Hobart TAS, Australia, <sup>5</sup>BRGM, Orléans, France, <sup>6</sup>Mediterranean Institute for Advanced Studies (IMEDEA), Palma, Spain, <sup>7</sup>Department of Physics, University of the Balearic Islands, Palma, Spain, <sup>8</sup>Department of Physics and Astronomy Augusto Righi, University of Bologna, Bologna, Italy, <sup>9</sup>Institute of Geography, Christian-Albrechts University Kiel, Kiel, Germany, <sup>10</sup>Institute for Environmental Studies (IVM), Vrije Universiteit Amsterdam, Amsterdam, Netherlands, <sup>11</sup>Deltares, Delft, Netherlands, <sup>12</sup>Tyndall Centre for Climate Change Research, University of East Anglia, Norwich, UK, <sup>13</sup>Institute for Marine and Atmospheric Research Utrecht and Department of Physical Geography, Utrecht University, Utrecht, Netherlands, <sup>14</sup>Department of Civil, Environmental and Construction Engineering, National Center for Integrated Coastal Research, University of Central Florida, Orlando, FL, USA

**Abstract** This study provides a literature-based comparative assessment of uncertainties and biases in global to world-regional scale assessments of current and future coastal flood risks, considering mean and extreme sea-level hazards, the propagation of these into the floodplain, people and coastal assets exposed, and their vulnerability. Globally, by far the largest bias is introduced by not considering human adaptation, which can lead to an overestimation of coastal flood risk in 2100 by up to factor 1300. But even when considering adaptation, uncertainties in how coastal societies will adapt to sea-level rise dominate with a factor of up to 27 all other uncertainties. Other large uncertainties that have been quantified globally are associated with socio-economic development (factors 2.3-5.8), digital elevation data (factors 1.2–3.8), ice sheet models (factor 1.6–3.8) and greenhouse gas emissions (factors 1.6–2.1). Local uncertainties that stand out but have not been quantified globally, relate to depth-damage functions, defense failure mechanisms, surge and wave heights in areas affected by tropical cyclones (in particular for large return periods), as well as nearshore interactions between mean sea-levels, storm surges, tides and waves. Advancing the state-of-the-art requires analyzing and reporting more comprehensively on underlying uncertainties, including those in data, methods and adaptation scenarios. Epistemic uncertainties in digital elevation, coastal protection levels and depth-damage functions would be best reduced through open community-based efforts, in which many scholars work together in collecting and validating these data.

**Plain Language Summary** One of the main impacts of climate change is sea-level rise leading to more frequent flooding of low lying coastal areas through higher tides, storm surges and waves. In this context, assessments of current and future coastal flood risk at global to world-regional scales are needed to inform policy decisions on greenhouse gas reduction targets and finance of adaptation and flood disaster risk reduction. A key requirement for such assessments is that they consider all major uncertainties in models, methods and data applied, because the failure to do so may lead to poor policy outcomes. So far, this key requirement has not been met. To address this limitation, this paper provides the first comparative assessment of uncertainties in global to world-regional scale studies of current and future coastal flood risks based on the published literature. We find that globally, by far the largest uncertainty concerns how coastal societies will adapt to sea-level rise, which can influence future flood risk by factors 20–27. Other large global uncertainties are associated with socio-economic development, digital elevation data, greenhouse gas emissions, and ice sheet evolution, influencing global exposure and flood risk by factors of up to 2 to 6.



# 1. Introduction

The increase of damages due to flooding caused by coastal extreme sea-level events, resulting from the interplay of tides, mean sea level rise, storm surges, and waves, may be one of the costliest aspects of climate change. Global to world-regional scale (called broad scale, hereafter) assessments of current and future coastal flood risks (CFR) are thus needed to inform a range of policy decisions including: (i) setting global mitigation targets in the context of the United Nations Framework Convention on Climate Change (UN-FCCC) to avoid "dangerous interference with the climate system" (UNFCCC, 1992); (ii) informing Global Assessment Reports on Disaster Risk Reduction by the United Nations Office for Disaster Risk Reduction (UNDRR, 2019); (iii) designing global financial mechanisms for adaptation (UNEP, 2016), disaster relief and loss & damage (Jongman et al., 2014); and (iv) strategic long-term development and adaptation planning.

A key requirement for informing these policy decisions, as well as for informing decisions in general, is that underlying assessments need to consider all major uncertainties in models, methods and data applied, because the failure to do so may mislead policy decisions leading to poor policy outcomes (Jones et al., 2014; Kunreuther et al., 2013; Morgan et al., 1990; Simpson et al., 2016).

The state-of-the-art of climate impact assessments generally, and broad-scale CFR assessments specifically, does not yet meet this requirement. Most efforts, notably those under the Climate Model Intercomparison Project (Eyring et al., 2016) and the Intergovernmental Panel on Climate Change (IPCC, 2014a), have focused on the exploration of uncertainty in climate models under different emission scenarios. More recently, a range of impact model intercomparison projects united under the umbrella of the Inter-Sectoral Impact Model Intercomparison Project (ISIMIP; https://www.isimip.org/) have started to explore uncertainties in impact models across different sectors such as water (Zaherpour et al., 2018) or forests (Petter et al., 2020). However, there are also many uncertainties beyond climate and impact models, which have been hardly explored, even though many of these uncertainties are known to be substantial. For broad-scale CFR assessments, these include uncertainties in subnational population data (Merkens et al, 2016), flood depth damage functions (Huizinga et al., 2017), extreme value analysis (EVA) methods (Mentaschi et al., 2016; Wahl et al., 2017), defense failure mechanisms (Allsop et al., 2007), and adaptation scenarios. To the best of our knowledge, no study has attempted to assemble and compare all major dimensions of uncertainty underlying current and future CFR.

This study contributes to filling these gaps by providing a literature-based comparative assessment of the major sources of uncertainty and biases relevant in broad-scale CFR assessments. We focus on broad-scale assessments, because the methods applied differ from methods applied in local-scale assessments (de Moel et al., 2015), mainly due to the limited availability of global data and computational resources. Assessing and comparing uncertainties in these methods is particularly timely, because they have developed substantially in recent years (Abadie et al., 2016; Diaz, 2016; Hallegatte et al., 2013; Hinkel et al., 2014; Lincke & Hinkel, 2018; Tiggeloven et al., 2020; Voudoukas, Mentaschi, et al., 2020; Vousdoukas et al., 2018) with broad-scale extreme sea-level models and datasets becoming increasingly available (Calafat & Marcos, 2020; Mentaschi et al., 2017; Morim et al., 2019; Muis et al., 2020; Tadesse et al., 2020; Vitousek et al., 2017; Vousdoukas et al., 2017; Vousdoukas et al., 2016).

In our uncertainty assessment we consider uncertainty in drivers and future projections of the four components of CFR, following the risk definition of the Intergovernmental Panel on Climate Change (IPCC) (Oppenheimer et al., 2019; Wong et al., 2014):

- 1. *Mean and extreme sea-level hazards*, including sea-level rise, tides, surges, waves, river run-off and their interactions;
- 2. *Hazard propagation* onto the shore and the floodplain, including interaction with natural (e.g., dunes) and artificial (e.g., dikes) defences;
- 3. Exposure in terms of area, people and coastal assets potentially threatened by these hazards; and
- 4. *Vulnerability*, which refers to the propensity of the exposure to be adversely affected by the flood hazard (IPCC, 2014b).

For each component and driver we extract, to the extent available in the literature, quantitative estimates of how sensitive components and resulting flood risk are to variations in the drivers. Finally, we compare





Figure 1. Methodological steps in broad-scale coastal flood risk assessments. Green boxes denote scenarios, blue boxes methods and yellow boxes data. Abbreviations: ESL=extreme sea-levels, MSL=mean sea-level, GIS = Geographic Information Systems, SLR=sea-level rise, EVA= extreme value analysis.

results across components and drivers, and provide directions for future research toward the goal of attaining broad-scale CFR estimates that consider all major dimensions of uncertainty and thus adequately inform relevant policy processes. This paper is a product of the Coastal Impact Model Intercomparison Project (COASTMIP; www.coastmip.org), a community-driven effort bringing together coastal system and impact modellers from around the world to better understand and project the long-term impacts of climate change on coastal systems.

# 2. Materials and Methods

Broad-scale assessment of CFR involves the application of many datasets and chains of numerical and statistical models, including climate models, land-ice models, tide, surge and waves models, defense failure models, inundation models, and damages models. Figure 1 provides an overview of how these data and models are generally combined in CFR assessments. Each step involved is discussed in more detail in the subsections that follow. A major methodological difference in those broad-scale CFR assessments is that they have either been conducted for collections of major European or global coastal cities (Abadie, 2018; Abadie et al., 2016; Hallegatte et al., 2013; Hunter et al., 2017; Prahl et al., 2018) or for entire coasts at continental or global scales (Brown et al., 2016, 2019; Diaz, 2016; Hinkel et al., 2014; Lincke & Hinkel, 2018; Nicholls et al., 2018; Tiggeloven et al., 2020; Vousdoukas et al., 2020). While the former studies generally take information on extreme sea-levels directly from observations (e.g., tide gauges located near the cities), the latter ones require the application of tide, surge and wave models to also have information for ungauged locations.

In an ideal situation, uncertainty assessment could proceed as a *global sensitivity analysis* using an integrated modeling system that covers all of the steps of Figure 1, and allowing all uncertain variables to vary



Uncertainty in Global Mean Sea-Level Rise. A->B Denotes a Variation in a Variable From Value A to Value B and the Variation Factor is B/A

	Variation in source variable					
Variable	Variation	Study	Assumptions	Variation (cm)	Factor	
Emissions (mean SLR forcing)	RCP2.6 -> RCP8.5	(Oppenheimer et al., 2019)	17th percentile	29->61	2.1	
			50th percentile	43->84	2.0	
			83rd percentile	59->110	1.9	
	Low (slightly warmer than RCP2.6)	(Bamber et al., 2019)	17th percentile 36->62	1.7		
	-> High (roughly RCP8.5)		50th percentile	69->111	1.6	
			83rd percentile	126->238	1.9	
Climate and land ice	17th -> 83rd percentile(Oppenhol17th -> 83rd percentile	(Oppenheimer et al., 2019)	RCP2.6	29->59	2.0	
models			RCP8.5	61->110	1.8	
	17th -> 83rd percentile	17th -> 83rd percentile (Bamber et al., 2019)	RCP2.6 <sup>a</sup>	49->98	2.0	
	17th -> 83rd percentile 5th -> 95th percentile		RCP8.5 <sup>a</sup>	79->174	2.2	
			RCP8.5 <sup>a</sup>	62->238	3.8	
Ice sheet models	Process-model based -> expert judgment based ice sheet contributions	Oppenheimer et al. (2019)	RCP2.6 <sup>a</sup> , 83rd percentile	59->98	1.7	
		-> Bamber et al. (2019)	RCP8.5 <sup>a</sup> , 83rd percentile	110->174	1.6	
Climate models	Lowest -> highest climate model (4	Hinkel et al. (2014)	RCP2.6, 50th percentile	30->39	1.3	
	models)		RCP8.5, 50th percentile	73->86	1.2	

<sup>a</sup>Bamber et al. (2019) do not use RCP8.5, but a comparable scenario that stabilizes global mean temperature at  $+5^{\circ}$ C in 2100 relative to 1850–1900. The median global warming under RCP8.5 is  $+4.3^{\circ}$ C in 2081–2100 relative to 1850–1900 (Collins et al., 2013). Likewise they do not use RCP2.6, but a comparable scenario that stabilizes global mean temperature at  $+2^{\circ}$ C in 2100 relative to 1850–1900. The median global warming of RCP2.6 is  $+1.6^{\circ}$ C in 2081–2100 relative to 1850–1900. The median global warming of RCP2.6 is  $+1.6^{\circ}$ C in 2081–2100 relative to 1850–1900 (Collins et al., 2013).

simultaneously (Saltelli et al., 2008). Given the number of datasets, models (including their alternative formulations and parameterizations), and the large number of uncertain variables involved, as well as the high computational costs required for running these models, this is far from being possible today. As a consequence, the available literature on uncertainty in CFR specifically, and on climate impacts generally, has focused on exploring few selected dimensions of uncertainty, mostly by varying one or few uncertain variables at a time (Frieler et al., 2017). Furthermore, the literature is compartmentalized into sets of literature addressing uncertainty in individual components of CFR. For example, part of the literature focuses on mean-sea-level rise uncertainty, part focuses on extreme sea-level uncertainty, part focuses on wave uncertainty, and part focuses on uncertainty in flood exposure. As our paper assesses uncertainty based on the published literature, we structure the presentations of results according to these components.

For each of the four components of CFR we consider one or several *target variables*, which are the outcome variables that broad-scale studies generally report upon. These are mean-sea-level rise (Table 1), extreme sea-levels (Table 2), wave heights (Table 3), flood damages (Tables 4 and 6), and area, people and asset exposure (Table 5). For each of the target variables, we consider one or several *sources of uncertainty* pertaining to the following three dimensions:

- 1. *Scenario uncertainties*, which are due to unpredictable human choice and include here socio-economic development scenarios, greenhouse gas emission/concentration scenarios and adaptation scenarios.
- 2. *Epistemic uncertainties*, which are due to imperfect knowledge and hence can be reduced in principle. This includes data uncertainty (e.g., digital elevation data, population data, etc.), climate model uncertainty (including downscaling methods), impact model uncertainty (e.g., hydrodynamic model used to simulate tides, waves and surges and their interactions; defense failure models and inundation models applied) and methodological uncertainty (e.g., uncertainty in methods for extreme value analysis).
- 3. *Aleatory uncertainty*, which is internal to the system studied and cannot be reduced (e.g., natural climate variability).



Uncertainty and Bias in Current and Future Extreme Sea-Levels. A->B Denotes a Variation in a Variable From Value A to Value B and the Variation Factor is B/A

Variation of source variable			Effect on extreme sea level (ESL)				
Variable	Study area	Variation	Assumptions (study)	Variable	Spatial metric <sup>a</sup>	Variation (cm)	Factor
Current extreme se	ea-levels (ESL)						
Tide-surge	Global tide	GTSR model -> tide gauge	(Wahl et al. 2017)	100-year ESL	Mean	+33	NR
modeling	gauges	observations	(Muis et al., 2017)	100-year ESL Mean		+19	NR
	European tide gauges	Statistical model -> tide gauge observations	(Calafat & Marcos, 2020)	50-year ESL	Median	±10	NR
	Global tide	Including statistical dependence	(Arns et al., 2020)	100-year ESL	Mean	+28	NR
	gauges	between tides and surges -> not including this			Max	+70	1.3
Wave set up	Global coastline	No wave set up -> wave set-up (0.2*Hs)	(Kirezci et al., 2020)	100-year ESL	Max	+50	NR
		Wave set-up w/o -> wave setup with	(Marcos et al., 2019)	50-year ESL	Median	+91	NR
		accounting for interdependence between surges and waves			Max	+300	NR
EVA methods	Global coastline	Generalized Pareto Distribution- >Gumbel distribution	(Wahl et al. 2017)	100-year ESL	Mean	+22	NR
	Global tide-gauges	Decadal variability in global tide gauge records	(Marcos et al., 2015)	50-year ESL	Mean	±5-6	NR
ESL data record	Global coastline	20 years -> 70 years record length	(Wahl et al. 2017)	100-year ESL	Mean	+15	NR
	Northern Australia coast	61 years climate simulation -> 10,000 years of synthetic tropical cyclones	(Haigh et al., 2014)	1,000-year ESL	Max	300-> 600	2.0
Extreme sea-levels	in 2100						
Emissions	E Africa	Constant storminess -> considering	RCP8.5 (Vousdoukas	100-year ESL	Median	-21	0.8
(atmospheric forcing) E China Sea/Sea of Japan		changes in storminess	et al., 2018)		Median	+27	NR
	North Europe		RCP8.5 (Vousdoukas	100-year ESL	Median	+6	1.1
	South Europe		et al, 2017)		Median	-20	NR
Wave set up	Global coastline	W/o wave set-up -> with wave set up	RCP8.5 (Kirezci et al., 2020)	100-year floodplain (10 <sup>3</sup> km <sup>2</sup> )	Mean	780 -> 820	1.1
Climate models	European coastline	Ensemble spread among six CMIP5 climate models	(Vousdoukas et al, 2017)	100-year ESL	Mean	NR	1.2
EVA methods	Global tide gauges	Stationary -> instationary EVA methods	(Vousdoukas et al., 2017; Mentaschi et al., 2016)	100-year ESL	Mean	+10	1.1

*Note.* NR Stands for Values Not Reported and AIS for Antarctic Ice Sheet.

<sup>a</sup>Spatial metric denotes the method applied to aggregate values from different locations (e.g., grid cells, tide gauges) of the study area. This includes "Mean", "Median", "Max" (i.e., maximum) and, in the case results are reported for a single location, "None".

We assess by how much the target variables vary within the uncertainty range of each individual source variable. The results are shown in Tables 1–6. If possible, we report on uncertainty ranges of target variables in absolute and relative terms. We thereby denote the absolute variation in a variable from value A to value B as "A->B". If a study does not report on A and B explicitly, we report on the absolute difference between A and B (i.e., B–A). The relative size of the variation in the target variable is given as the *variation factor* B/A. This factor is then used to compare uncertainties across target and source variables.

To the extent allowed by the published studies, we try to use consistent uncertainty ranges for the source variables. For greenhouse gas emission/concentration uncertainty, we use the range from the representative



Uncertainty and Bias in Current and Future Wave Height. A->B Denotes a Variation in a Variable From Value A to Value B and the Variation Factor is B/A

	Variation in source variable			Effect on wave height			
Source	Study area	Variation	Assumptions (study)	Variable	Spatial Metric <sup>a</sup>	Variation (cm)	Factor
Current waves							
Wave models	Global	Observation -> model	Ensemble of wave models (Morim et al., 2019)	CRMSD <sup>b</sup> of mean Hs <sup>c</sup>	Max	+50	NR
		Lowest-> highest calibration wave data set	WWIII, 12 wave datasets used (Stopa, 2018)	100-year Hs <sup>c</sup>	Max	900->1,700	1.7
	North Atlantic	WWIII -> WAM	Same wind forcing (Swain et al., 2017)	Mean wave height	Mean	NR	1.1
Wave model resolution	Global	40 km -> 150 km spatial resolution	(Mentaschi, 2018)	10-year Hmax <sup>d</sup>	Mean	NR	0.96
	Area with tropical cyclones	1° -> 0.25° spatial resolution	(Timmermans et al., 2017)	Hmax <sup>d</sup>	Max	+1,500	NR
	Areas w/o tropical cyclones				Max	+500	NR
EVA methods	Global	EVA on single model -> EVA on pooled data from model ensembles	(Meucci et al., 2018)	100-year Hs <sup>d</sup>	Max	1,400-> 1,600	1.1
Future waves (year 2	2100)						
Emissions (atmospheric forcing)	Global	Today -> 2100	Climate and wave mode ensembles, RCP4.5 and 8.5 (Morim et al., 2019)	Mean Hs <sup>b</sup>	Mean	NR	1.1–1.2
	Central North	Today -> 2100	RCP4.5 (Aarnes et al., 2017)	Mean wave	Mean	NR	0.94
	Atlantic	Today -> 2100	RCP8.5 (Aarnes et al., 2017)	height	Mean	NR	0.90
		RCP4.5 -> RCP8.5	(Aarnes et al., 2017)	Mean wave height	Mean	NR	0.96
Climate models	Global	Lowest -> highest climate	RCP8.5 (Morim et al., 2019)	Mean Hs <sup>b</sup>	Max	NR	1.2
(atmospheric forcing)		and wave model combination		100-year Hs <sup>b</sup>	Max	NR	1.4
EVA methods	Global	EVA on single model -> EVA on pooled data from model ensembles	(Meucci et al., 2020)	100-year Hs <sup>b</sup>	Max	+500	1.5

Note. NR stands for values not reported.

<sup>a</sup>Spatial metric denotes the method applied to aggregate values from different locations (e.g., grid cells, tide gauges) of the study area. This includes "Mean", "Median", "Max" (i.e., maximum) and in the case results are reported for a single location, "None". <sup>b</sup>centered-root-mean-square-difference. <sup>c</sup>significant wave height.

concentration pathways (RCP) 2.6 to 8.5, as this is the range most commonly reported upon in the literature. For socioeconomic uncertainty we use the range over the Shared Socioeconomic Pathways (SSP), which is the standard set of socioeconomic scenarios used in climate change-related research and consists of five alternative futures describing different challenges to adaptation and mitigation (Kriegler et al, 2012; O'Neill et al., 2017). We acknowledge that these two ranges might not necessarily span the full range of uncertainties. For example, alternative socio-economic scenarios have come up with both higher and lower population numbers in 2100 than the SSPs (Vollset et al., 2020). Similarly, some authors argue that there is a 35% chance of exceeding RCP8.5 (Christensen et al., 2018) and others argue that RCP8.5 is an extreme and very unlikely case (Hausfather & Peters, 2020).

Generally, we aim to report on both uncertainty and bias. Bias is assessed either as the difference between observations and model or method results. When observations are not available, as is the case for many



Uncertainty and Bias in Flood Propagation. A->B Denotes a Variation in a Variable From Value A to Value B and the Variation Factor is B/A

Variation in source variable			Effect on flood damage			
Variable	Study area	Variation	Assumptions (study)	Variable	Variation	Factor
Current flooding						
Inundation model	Europe	LISFLOOD-FP -> Bathtub	(Vousdoukas et al., 2016)	Flood area (10 <sup>3</sup> km <sup>2</sup> )	31->50	1.6
Future flooding (2100)						
SLR scenario	Global	RCP2.6 -> RCP8.5	Robust CBA adaptation, SPP2 (Lincke and Hinkel, 2018)	EAD <sup>a</sup> (billion US\$/ yr))	230-> 420	1.8
Socio-economic scenario	Global	SSP3 (lowest scenario) -> SSP5 (highest scenario)	Robust CBA adaptation, RCP8.5 (Lincke and Hinkel, 2018)	EAD <sup>a</sup> (billion US\$/ yr)	260-> 590	2.3
			Constant protection levels, RCP8.5 (Tiggeloven et al., 2020)	EAD <sup>a</sup> (billion US\$/ yr)	<sup>b</sup> 380-> 2,200	<sup>b</sup> 5.8
Adaptation scenario	Global	Demand for safety protection -> no adaptation	3 RCPs, 5 SSP (Hinkel et al., 2014)	Max. EAD <sup>a</sup> (billion US\$/yr)	75-> 100,000	1300.0
		Cost-benefit protection -> constant protection levels	RCP2.6, SSP2 (Nicholls et al., 2019)	EAD <sup>a</sup> (billion US\$/ yr)	55-1,1000	20.0
			RCP8.5, SSP2 (Nicholls et al., 2019)	EAD <sup>a</sup> (billion US\$/ yr)	120->3,200	26.7

<sup>a</sup>EAD= expected annual damages. <sup>b</sup>These are 2080 values as the authors do not report on 2100.

components of CFR, we report on uncertainty ranges for target variables obtained through the use of alternative datasets, models or methods, as generally done in climate and impact model intercomparison projects. Uncertainties in future components of flood risk are reported for 2100, because this is the time horizon most frequently used in assessments.

## 3. Results

#### 3.1. Hazard

#### 3.1.1. Mean Sea-Levels

*Methods applied*. The mean sea-level metric relevant for CFR assessments is local relative sea-level change, which is generally obtained by combining information on the following components: (i) sterodynamic SLR obtained from multi-model ensembles of Atmosphere-Ocean General Circulation Models (AOGCM), (ii) contribution of land-ice models (Antarctica, Greenland) and glaciers forced by AOGCM output in terms of mainly temperature and precipitation, (iii) contribution of land water storage changes. From these components local relative sea-level rise is obtained with a sea-level equation model that calculates the gravitational and rotational patterns in sea-level rise (Slangen et al., 2014); (iv) contribution of glacial-isostatic adjustment; and v) contribution of uplift and subsidence processes, especially in geologically recent sedimentary deposits such as deltas and alluvial plains (Nicholls et al., 2014). Results are typically expressed by their mean values and a selected percentiles (e.g., 17%–83%, Oppenheimer et al., 2019), but full probability density distributions have also been produced (Kopp et al., 2014).

*Major uncertainties*. According to the process-model based assessment of the IPCC's Special Report on the Oceans and the Cryosphere in a Changing Climate (SROCC; Oppenheimer et al., 2019), uncertainties in 21st century global mean SLR due to uncertainty in emission scenarios and uncertainty in models (climate and land-ice) are roughly at equal footing (Table 1). The minor part of model uncertainty relates to climate model uncertainty, which has been found to influence global mean SLR in 2100 by factors 1.2–1.3 (i.e., 13 cm under RCP8.5) for an ensemble of 4 climate models (Table 1; Hinkel et al., 2014). Considering a larger ensemble of climate models, Little et al. (2015) find a variation of steric global mean SLR of about 20 cm under RCP8.5 in 2090, but the authors do not report on total global mean SLR. In any case, the major part of model uncertainty is associated with the land-ice contributions of Greenland and Antarctica to global



Uncertainty in Current and Future Exposure. A->B Denotes a Variation in a Variable From Value A to Value B and the Variation Factor is B/A

Variation in source variable		Effect on exposure				
Variable	Study area	Variation	Assumptions (study)	Variable	Variation	Relative
Current exposure						
Elevation datasets	Global	SRTM -> GLOBE	(Lichter et al., 2011)	LECZ <sup>a</sup> area (10 <sup>3</sup> km <sup>2</sup> )	2,400-> 2,800	1.2
		SRTM90 -> CoastalDEM	Landscan-2010 (Kulp & Strauss, 2019)	LECZ <sup>a</sup> population (millions)	780->1040	1.3
		SRTM -> GLOBE	GRUMP (Hinkel et al., 2014)	100-year floodplain (10 <sup>3</sup> km²)	660-> 1,200	1.8
				100-year floodplain population (millions)	160 -> 310	1.9
	SRTM90 -> CoastalDEMLandScan-2010 (Kulp & 1-year floodplain population Strauss, 2018)1-year floodplain population (millions)		1-year floodplain population (millions)	65 -> 250	3.8	
		SRTM90 -> GLL_DTM_v1	Vernimmen et al., (2020)	Land below 2 m between 60°N and 56°S (10 <sup>3</sup> km <sup>2</sup> )	317->934	2.9
Population datasets	Global	LandScan -> GRUMP	SRTM (Hinkel et al., 2014)	100-year floodplain	93->160	1.7
		population data	GLOBE (Hinkel et al., 2014)	population (million)	290 -> 310	1.1
Future exposure (2100)						
SLR scenarios	Global	RCP2.6 -> RCP8.5	SSP2; 50th percentile SLR (Merkens et al., 2018)	100-year floodplain population (millions)	149->170	1.1
		17th percentile -> 83rd percentile SLR	RCP8.5; SSP2 (Merkens et al., 2018)		155->187	1.2
		Without -> with wave set up	RCP8.5 (Kirezci et al., 2020)	100-year floodplain (10 <sup>3</sup> km²)	780 -> 820	1.1
Socio-economic	Global	Lowest -> highest SSP	(Merkens et al., 2016)	LECZ population (millions)	850 -> 1,200	1.4
scenarios			(Jones & O' Neill, 2016)		490 -> 1,100	1.5
		National -> regionalized	SSP1 (Merkens et al., 2016)	LECZ population (millions)	620 -> 850	1.4
		population projections	SSP5; 50th percentile SLR; RCP6.0 (Merkens et al., 2018)	100-year floodplain population (millions)	129 -> 179	1.4

Note. NR stands for values not reported.

<sup>a</sup>low elevation coastal zone.

mean SLR (van de Wal et al., 2019). Uncertainties in the ice-sheet contributions are high because some of the processes that may lead to large contributions by the end of the century are only captured in a highly parameterized way in state-of-the-art ice-sheet models. This includes hydrofracturing of ice shelves, leading to enhanced mass loss from the ice sheet by marine ice sheet or marine ice cliff instability (DeConto & Pollard, 2016; Pattyn et al., 2018), or the development of dark ice surfaces due to inorganic matter or ice algae accumulating due to warmer and wetter conditions on the ice sheet of Greenland, which in turn accelerates surface melting (Tedstone et al., 2017). For this, and other reasons, expert elicitation studies (Bamber et al., 2019) on the contributions of the ice-sheets to SLR consistently produce higher SLR estimates than process-model based studies, particularly for higher RCPs and higher percentiles. In these studies, 21st century global mean SLR is about twice as sensitive to model uncertainties (including expert judgment for ice-sheets) as compared to emission uncertainties (Table 1) for the 95% percentile.

Related to this, epistemic uncertainties exist on the covariance between the contributions of different components (Lambert et al., 2021). Most studies use Monte-Carlo Analysis assuming complete independence, which certainly is not justified for some components as they are driven by the same climate forcing. Including this could significantly increase or decrease the uncertainty in the local relative SLR and thereby affect the higher and lower percentiles of the probability distribution function.



Uncertainty in Current Local Vulnerability and Flood Damage. A->B Denotes a Variation in a Variable From Value A to Value B and the Variation Factor is B/A

Variation in source variable					
Study area	Variation	Assumptions (study)	Variable	Variation	Factor
The Netherlands	2.5th -> 97.5th percentile using alternative damage models	(de Moel et al., 2012)	Damage	NR	8.0
Eilenberg (Germany)	Lowest -> highest damages of 7 different river flood damage models	(Jongman et al., 2012)	Damage (million €)	55-> 470	8.5
Rotterdam	No freeboard -> 50 cm freeboard	(de Moel et al., 2013)	Damage	NR	0.5
New York	local DDF (Aerts et al, 2014) -> continental DDF	(Huizinga et al., 2017)	Damage (billion €)	44->99	2.3
São Tomé and Príncipe	Local DDF -> country/continental scale DDF	(Parodi et al., 2020)	Damage	NR	16.0
	Study area Study area The Netherlands Eilenberg (Germany) Rotterdam New York São Tomé and Príncipe	Variation in source variableStudy areaVariationThe Netherlands2.5th -> 97.5th percentile using alternative damage modelsEilenberg (Germany)Lowest -> highest damages of 7 different river flood damage modelsRotterdamNo freeboard -> 50 cm freeboardNew Yorklocal DDF (Aerts et al, 2014) -> continental DDFSão Tomé and PríncipeLocal DDF -> country/continental scale DDF	Variation in source variableStudy areaVariationAssumptions (study)The Netherlands2.5th -> 97.5th percentile using alternative damage models(de Moel et al., 2012)Eilenberg (Germany)Lowest -> highest damages of 7 different river flood damage models(Jongman et al., 2012)RotterdamNo freeboard -> 50 cm freeboard(de Moel et al., 2012)New Yorklocal DDF (Aerts et al, 2014) -> continental DDF(Huizinga et al., 2017)São Tomé and PríncipeLocal DDF -> country/continental scale DDF(Parodi et al., 2020)	Variation in source variable    Variation      Study area    Variation    Assumptions (study)    Variable      The Netherlands    2.5th -> 97.5th percentile using alternative damage models    (de Moel et al., 2012)    Damage      Eilenberg (Germany)    Lowest -> highest damages of 7 different river flood damage models    (Jongman et al., 2012)    Damage (million €)      Rotterdam    No freeboard -> 50 cm freeboard    (de Moel et al., 2013)    Damage et al., 2013)      New York    local DDF (Aerts et al, 2014) -> continental DDF    (Huizinga et al., 2017)    Damage (billion €)      São Tomé and Príncipe    Local DDF -> country/continental scale DDF    (Parodi et al., 2020)    Damage	Variation in source variable    Variation    Variation in flood damage      Study area    Variation    Assumptions (study)    Variable    Variation      The Netherlands    2.5th -> 97.5th percentile using alternative damage models    (de Moel et al., 2012)    Damage    NR      Eilenberg (Germany)    Lowest -> highest damages of 7 different river flood damage models    (Jongman et al., 2012)    Damage (million €)    55-> 470      Rotterdam    No freeboard -> 50 cm freeboard    (de Moel et al., 2013)    Damage    NR      New York    local DDF (Aerts et al, 2014) -> continental DDF    (Huizinga et al., 2017)    Damage (billion €)    44->99      São Tomé and Príncipe    Local DDF -> country/continental scale DDF    (Parodi et al., 2020)    Damage    NR

Note. NR stands for values not reported.

Another major dimension of local mean sea-level uncertainty is human-induced subsidence, which is mainly the enhancement of sediment compaction through the actions of humans, especially through groundwater withdrawal (Shirzaei et al., 2020; Syvitski et al., 2009). The effect of this is small in terms of global mean sea-levels, but it is large in terms of flood risk, because coastal populations are preferentially located in subsiding locations such as the large river deltas and their associated cities in Asia. While the global average rate of mean sea-level rise is 3.3 mm/yr (Oppenheimer et al., 2019), the average rate experienced in subsiding regions is currently up to three times faster at 8–10 mm/yr (Nicholls et al., 2021). Maximum current subsidence rates in some of the worst affected delta cities are as high as 120 mm/yr in Bangkok and 180 mm/yr in Jakarta (Erkens et al., 2015; Herrera-García et al., 2021). These rates are, however, difficult to extrapolate into the future, because even such high rates of subsidence can be reduced or stopped through appropriate water management measures, as has happened, for example, in Tokyo. The global effects of this uncertainty have therefore not been explored yet.

## 3.1.2. Extreme Sea-Levels

*Methods applied*. Global data on ESL hazard is generated based on high frequency tide gauge records where available, and on numerical tidal, storm surge, wave and river models for ungauged locations and future conditions. Numerical storm surge simulations generated with process-based models are now becoming available at broad scales (Muis et al., 2020; Vousdoukas et al., 2017). Computationally more efficient numerical models based on statistical relationships between surges and atmospheric pressure fields have demonstrated similar performance as process-based models, at least in some regions (Cid et al., 2018; Rashid & Wahl, 2020; Tadesse et al., 2020), but they require observations (or output from process-based models) for training and have not yet been applied for broad-scale CFR assessments. A new observation-based probabilistic assessment of ESL has recently been applied along the European coasts (Calafat & Marcos, 2020), providing improved accuracy at both gauged and ungauged sites. Despite its promising performance, these types of models are still at their initial stages and further developments are needed for CFR.

*Tide-surge models* generally underestimate observed ESL by a few decimeters for the 100 years events on average (Table 2), but the underestimation of the surge models of the strongest events is much higher than the global average value. Differences between modeled and observed ESL are significantly larger in areas hit by tropical cyclones, specifically for large return periods, because the temporal and spatial resolution of climate reanalysis/simulations are insufficient to fully include the strong winds of tropical cyclones and do not contain a sufficient number of tropical cyclones to obtain reliable statistics of extreme values (Hodges et al., 2017; Muis et al., 2020; Woodruff et al., 2013). For example, maximum observed ESL in New Orleans during hurricane Katrina have been found to be 2.7 to 4 times higher than simulated 1000-yr ESL (Muis et al., 2016). Ongoing work is addressing these limitations. For example, using high-resolution climate data (Bloemendaal et al., 2018) or parametric wind models combined with best track data (Lin & Emanuel, 2016) are able to resolve the maximum surge height within centimeters (Haigh et al., 2014).

Furthermore, synthetic resampling techniques can be applied to extend the observed records to thousands of years (Bloemendaal et al., 2020; Emanuel et al., 2006).

*Wave set-up and run-up.* Wind-waves contribute to ESL via three processes: infragravity waves, wave setup and wave runup (Dodet et al., 2019) with uncertainties associated with each, introduced via uncertainties in: (i) offshore wave characteristics, that is, how well observed or simulated they are, which will be described in the next subsection; and (ii) how waves propagate and interact with the nearshore morphology and coastal profile. Considering wave contributions to ESL via process-based modeling is challenging at broad scales, due to the computational cost of large-scale numerical models with the necessary high spatio-temporal resolution, and a lack of observational records as coastal tide-gauges are generally preferentially positioned in locations sheltered from any contribution from wind-waves. As a result, only a few broad-scale CFR assessments have considered the contribution of wave-set up to coastal flooding, and those that have, used simple parameterisations dependent on offshore wave information and coastal morphology (Kirezci et al., 2020; Vousdoukas et al., 2018). Alternative parameterizations for assessing wave set up for sandy coasts are available, for example, setup =  $0.2^{*}$ Hs by Holthuijsen (2010), or that of Stockdon et al. (2006), which have been applied globally for beach shorelines (Melet et al., 2020; Rueda et al., 2017; Vitousek et al., 2017) or modified for other environments such as coral reefs (Beck et al., 2018).

The simple parameterization of Holthuijsen (2010) can significantly overestimate wave-setup locally. For example, using a local coupled surge-wave model, Amores et al. (2020) find a wave set-up of 40 cm caused by waves with a significant height of about 800 cm during the storm Gloria in the north-western Mediterranean, as compared to 160 cm that would be obtained by applying the Holthuijsen (2010). One major uncertainty in applying other parametrizations that rely on input parameters such as beach slope is the lack of broad-scale data on morphology across the shoreface (nearshore, foreshore and backshore), due to the lack of an observing system capable of measuring this at an affordable cost and appropriate spatio-temporal scales. To circumvent the problem, global studies have assumed a constant beach slope, for example, 0.1, by Melet et al. (2018), which can locally over- or under-estimate wave set-up given that observed beach slope ranges between 0.001 and 0.6 (10th to 90th percentiles) globally (Athanasiou et al., 2019). On a global average, the contribution of wave-set up to ESL is relatively small (Kirezci et al., 2020), but wave-set up can locally reach 40–50 cm under strong storm conditions (Amores et al., 2020; Bertin et al., 2015).

Wave-run up contributions were considered in broad scale studies by Melet et al. (2018, 2020). While these contributions are short time scale (on order of wind-wave frequencies) and unlikely to lead to sustained flooding, they can play an important role in initializing failure of coastal defenses such as dikes or dunes. Runup estimates are very sensitive to beach slope assumptions (Stockdon et al., 2006).

Statistical dependencies between ESL components. Further bias in broad-scale assessments of ESL is introduced through the non-consideration of the statistical dependence between surge, tide, river discharge and wave contributions to ESL. While tides are the major contributor to ESL globally (Merrifield et al., 2013), nonlinear tide-surge interactions are generally not considered in broad-scale assessments of ESL, which can overestimate ESL by up to 70 cm at some locations (Arns et al., 2020). Similarly, broad-scale ESL assessments generally do not include the influence of river discharge on ESL in river deltas and estuaries. This effect has not been quantified for river months at global scale, but it has been found that including the coastal ESL into river flood models increases extreme water levels on average by about 10 cm for many global deltas and estuaries (Eilander et al., 2020; Ikeuchi et al., 2017). Finally, the occurrence of high storm surges and wind waves is correlated at about 55% of the global coastline, and neglecting this effect can underestimate the contribution of wave setup to ESL significantly (Marcos et al., 2019). This also holds true for many ESL records from tide gauge locations, as gauges are usually located in wave-sheltered harbors and hence underestimate the contribution of waves to ESL (Lambert et al., 2020).

*Extreme value analysis (EVA) methods.* Uncertainties in ESL also arise from the different extreme value analysis (EVA) methods regarding the selected frequency analysis approach, statistical model applied and the return period curve fitting (Buchanan et al., 2017; Hamdi et al., 2014; Wahl et al., 2017). The Gumbel distribution, for example, which has been extensively used in broad-scale ESL studies (Hunter, 2012; Hunter et al., 2013; Muis et al., 2016), tends to overestimate global return levels by 22 cm on average as compared to the Generalized Pareto Distribution (Wahl et al., 2017). In many studies, stationary EVA models are applied

to quasi-stationary slices of data, typically with a length of 30 years (Vousdoukas et al., 2016). Nonstationary EVA methods enable studying time-varying return levels and thus increase the sample, generally leading to a decrease of the statistical uncertainty (Menendez et al., 2009; Mentaschi et al., 2016).

Furthermore, limitations in the observational data set, including short length of the time series, lack of representativeness and, associated to this, a lack of observed strong events, limits the accuracy of ESL estimates, in particular for long period return levels (Table 2). Globally, Wahl et al. (2017), for example, quantified that the 100-year ESL increases about 15 cm on global average, and up to more than half a meter at certain locations, when 70 years instead of 20 years of observations are used. This effect is specifically pronounced if exceptionally large extreme events have not been included in the extreme value analyses. Including such events, either by updating return levels after large events or by extending tide gauge records with information on ESL found in historical documents or through modeling efforts (see section on Tide-surge models) can reduce these uncertainties. For example, integrating historical records into the tide gauge record of Venice increases the 50-year ESL by factor 1.3 (Marcos et al., 2009). Decadal variability of 50-year ESL has been found to lie at around 10 cm (Marcos et al., 2015; Menendez & Woodworth, 2010; Rashid et al., 2019).

*Future ESL*. At broad-scales estimates of future ESL have been generally produced by considering the following two climate change effects separately: (i) effects of mean SLR on ESL (*mean SLR forcing*, hereafter) are captured by displacing ESL distributions upwards (or downwards) with changes in mean sea-levels, which means that the uncertainties involved are those related to mean sea-levels presented above; and ii) effects of changing atmospheric conditions on ESL (*atmospheric forcing* hereafter) are assessed by forcing surge models with wind and pressure data from climate models. The latter effect has been studied less at broad scales, but generally this effect is much smaller as compared to the former, estimated to influence ESL by less than 10% on global average under RCP4.5 and RCP8.5 (Vousdoukas et al., 2018). These estimates are median values based on multi-model ensembles and locally changes in storminess can have a larger effect on ESL, either positive or negative. Further uncertainties in estimating future ESL that have hardly been quantified at broad scales include ocean model errors related to the reduced spatial resolution of both the meteorological forcing and model grid (Calafat et al., 2014; Conte & Lionello, 2013).

Uncertainties related to *future tides*, which change due to a number of processes including mean sea-level change (Haigh et al., 2020) have generally not been considered in broad scale CFR analysis. SLR alters tides by reducing bottom friction, changing resonance properties and increasing reflection at the coast (Idier et al., 2017). The effect of SLR on tides has been estimated to be smaller than  $\pm 16$  cm change in MHW under 1 m SLR at the 136 largest coastal cities, assuming a fixed coastline (Pickering et al., 2017).

A further source of uncertainty in future ESL relates to *local nearshore effects of rising mean-sea levels on waves, tides and surges* Arns et al., 2015; Du et al., 2018; ;Roland et al., 2012; Schmitt et al., 2018; Zijl et al., 2013). For example, sea-level rise increases shallow water depth, which can increase tidal ranges and surges by reducing damping friction. Furthermore, deeper nearshore waters reduce wave set up but increase wave amplitudes and hence wave runup in the case of rigid/fixed coastlines (Cheon & Suh, 2016). In the case of sandy beaches that are able to retreat landwards wave setup will remain constant, since the beach profile will not change. These effects are specifically pronounced in shallow continental shelf areas such as the German Bight, where it has been found that these effects have the same order of magnitude as the direct increase of ESL through mean sea-levels (Arns et al., 2017). While some of these effects are beginning to be captured globally, for example, the effects of SLR on tides (Haigh et al., 2020) and tide-surge dynamics (Muis et al., 2020), these effects have so far not been considered in broad-scale CFR analysis.

*Combining MSL and ESL.* Finally, for the interpretation of projections of future ESL, it is important to note that different approaches have been applied for combining MSL and ESL. One approach adds ESL distribution to deterministic scenarios of MSL (Hallegatte et al., 2013; Hinkel et al., 2014) and the other approach convolutes ESL distributions with probabilistic sea-level scenarios (Buchanan et al., 2017; Vousdoukas et al., 2018). Care needs to be taken, because the resulting future ESL distributions have different interpretations. In the former case, the ESL distribution attained represents a possible ESL that could occur in the future under the assumption that the chosen deterministic SLR scenario materializes. In the latter case, the future distribution of ESL attained will never materialize (i.e., probabilistic scenarios do not materialize by



definition). The obtained distribution rather represents the likelihood of occurrence of a specific ESL at a given moment in the future.

## 3.1.3. Wind Waves

*Methods applied*. In-situ wave observations (wave buoys) provide measurements of the full wave spectrum in deep waters, thus resolving wave period and length. Wave buoys are, however, sparsely distributed globally, have limited record lengths and, in many instances, there are homogeneity issues due to changing buoy measuring platforms (Gemmrich et al., 2011). Globally wave observations have been available for the last 40 years from satellite altimetry, but this provides only information on wave heights at low temporal resolution (~10 days) and not on wave periods or directions. Hence, understanding global-scale wave characteristics relies heavily on numerical models, typically third-generation spectral models such as Wavewatch III (Tolman, 2009), WAM (Komen et al., 1996) or SWAN (Booij et al., 1997) or statistical models that capture the relationships between wave heights and atmospheric fields (Camus et al., 2011; Wang et al., 2014).

*Current wind waves*. Averaged over broad-scales, models simulate wave heights remarkably well, but variations in calibration data (observed waves) and forcing products (surface winds of varying spatial or temporal resolution) can lead to significant uncertainties (Table 3), typically greatest for the extreme waves, which are also those most relevant for flood risk. Low space-time resolution increases the uncertainty related to numerics in dynamical wave models and model calibrations can be resolution dependent. Furthermore, subscale processes such as unresolved islands can have significant consequences on the model skill if not properly parameterized (Mentaschi et al., 2020; Tolman, 2003). Locally, the uncertainty in extreme waves is thereby stronger for events dominated by mesoscale dynamics (Mentaschi et al., 2015), notably for tropical cyclones, for which increasing model resolutions can increase maximum wave heights significantly (Table 3). Similar to the surge component of sea-level hazard discussed above, model calibration and validation during tropical-cyclones is hampered by the scarcity of observations. Uncertainties in wave period and direction, which are equally important for the wave-related coastal flooding, are greater compared to wave heights.

Low spatial resolution of wave models is also a particular limitation for the coastal and nearshore zone, as the nearshore wave dynamics, such as wave setup, are poorly resolved (Saulter et al., 2017). Model resolution (spatial and spectral) also limits representation of important processes in determining wave driven coastal sea-level, for example, infragravity waves. Generally, the comparison between climate change wave studies is hampered by the inconsistency of wave variables reported (Morim et al., 2018).

*Future wind waves*. Broad scale ESL studies and CFR assessments commonly assume wave climate stationarity (Kirezci et al., 2020; Melet et al., 2018; Vitousek et al., 2017). Of studies that consider climate driven changes in wave characteristics, uncertainties in projections of future offshore wave conditions are dominated by uncertainties in future forcing from climate models, followed by wave model uncertainty and RCP uncertainty has the smallest contribution. Climate model uncertainty is globally significant and an order of magnitude larger than climate scenario uncertainty (Wang et al., 2015). Using an ensemble of 148 wave climate projections using different global climate and wave models shows robust changes in at least one parameter of wave climate (significant wave height, wave period and wave direction) for about 50% of the ocean (Morim et al., 2019). Under RCP4.5, however, all robust changes in wave climate fall within the present-day natural variability, but under RCP8.5 changes exceed this over ~50% of the world's ice-free ocean area. Other sources of uncertainty include unresolved forcing characteristics, for example, tropical cyclones (Appendini et al., 2017; Timmermans et al., 2017) and those associated with model resolution and uncertainties surrounding EVA methods.

Nearshore wave climate is generally more sensitive to climate change than offshore wave climate due to effects of SLR on coastal morphology (e.g., deeper waters) (Wandres et al., 2017). Uncertainties surrounding future coastal morphology including, for example, issues around sediment availability and shoreface slope changes (Cowell et al., 1995; Goodwin et al., 2006) and reef stability (Hongo et al., 2018) have, however, not been explored at broad-scale.





Figure 2. Defense failure mechanisms.

#### 3.2. Hazard Propagation

*Processes and methods applied.* The propagation of mean and extreme sea-levels into the hinterland causing coastal flooding is shaped by how sea levels interact with the coastal profile including the natural (e.g., dunes) and artificial (e.g., dikes, seawalls) flood barriers in place. The presence or absence of coastal protection and its design standard have large effects on flood extent and depth. If no barriers exist, ESL propagate inland where they exceed land elevation. Where barriers are present, flooding is caused by the following three *failure mechanisms*: (i) *defense overtopping* by waves, if wave runup exceeds the height of the defenses; (ii) *defense overflowing* by ESL, if ESL exceed the height of the defenses; and (iii) *defense breaching*, where part of the defense is removed by ESL and waves, or geotechnical failure (Figure 2). Furthermore, different types of inundation models are applied to assess flood propagation, ranging from static, bathtub approaches to hydrodynamic models.

*Defense failure mechanisms*. The uncertainties of many of the above processes have only been quantified at local scales (Le Cozannet et al., 2015). Modeling defense failure mechanisms, in particular wave overtopping and breaching, requires high resolution hydro-morphodynamic modeling and data on coastal profiles and defense designs, which are not available at broad scales. Hence broad scale CFR assessments have either: (i) focused on flood exposure and ignored coastal protection (Hanson et al., 2011); (ii) only considered overflow but not breaching (Vousdoukas Bianchi, et al., 2018); or (iii) only considered breaching assuming that once an ESL exceeds defense heights, defenses breach and fail completely (Diaz, 2016; Hinkel et al., 2014; Lincke & Hinkel, 2018; Tamura et al., 2019), or a combination of the latter two (Hallegatte et al., 2013). Ignoring coastal protection is misleading because extensive defense systems exist in developed and well populated coastal locations around the world, with notable concentrating on defense overflow considers that flood defenses may still provide some protection even if ESL exceeds their height, as defenses still reduce the amount of water flowing into the floodplain. However, it has also been found that coastal defenses often breach once overflow occurs (Hall et al., 2003). These different approaches have not been compared, and the uncertainties have not been assessed at broad-scales, but have been shown to be large at local scale. For the

Solent in the UK, for example, it has been shown that switching from simulating overflow to overtopping and breaching increases the number of properties inundated by factor 3.7 (Wadey et al., 2012). Not considering wave overtopping in broad-scale analysis leads to an underestimation of CFR in areas in which flooding is wave-dominated as found in mid to low latitudes such as western Australia, eastern Madagascar, the Maldives and small islands in the Pacific (Beetham & Kench, 2018; Rueda et al., 2017; Wadey et al., 2017).

*Current protection levels*. Independent of how defense failure mechanisms are modeled, there is a large uncertainty on current protection levels, because data on the presence of coastal defences, their nominal protection standard, probability of failure, and maintenance level are not systematically available at broad scales. While efforts are underway to collect some of this data (e.g., the FLOPROS database by Scussolini et al., 2016), expert judgment and modeling is presently required to fill the large gaps. For example, Yohe and Tol (2002) and Hinkel et al. (2014) model protection standards as a function of societal wealth and land use/population density. For the 136 largest coastal cities in the world using data supplemented by expert judgment it was estimated that 60% of city defences are below 1 in 100 years, 30% are 1-in-100 years, and 10 per cent were above 1-in-100 years up to 1 in 10,000 years for Amsterdam and Rotterdam in the Netherlands, where the highest defense standards in the world are presently found (Hallegatte et al., 2013). Using the FLOPROS modeling approach, Tiggeloven et al. (2020) have also estimated coastal flood protection for all regions of the world. Uncertainty in resulting protection levels differs substantially between regions but this has hardly been explored.

Inundation modeling. Local scale process-based inundation models are computationally too expensive for broad-scale analysis and require high resolution topographic data that are not readily available. Therefore, broad-scale studies have applied either computationally more efficient reduced-complexity models like LISFLOOD-FP (Bates et al., 2010), for example, at European scales (Vousdoukas et al., 2016), or a static inundation model (i.e., a bathtub approach) in which the coastal water levels are projected inland across the floodplain where defences are overtopped (Diaz, 2016; Hallegatte et al., 2013; Hinkel et al., 2014; Lincke & Hinkel, 2018; Tamura et al., 2019). Locally, the static approach has been found to overestimate flood extents in flatter terrains as compared to dynamic approaches by factor 0.5-2, when the main flooding process involved is overflow (Breilh et al., 2013; Gallien, 2016; Ramirez et al., 2016; Seenath et al., 2016). At broad scales, this has hardly been assessed. Only one study has compared the bathtub approach with LISFLOOD at the European scale, finding that flood extents using the former are about 1.6 times larger than using the latter (Vousdoukas et al., 2016). In this context it should be noted that hydrodynamic models are not necessarily providing better results either, but need to be calibrated to regional circumstances, which is difficult at broad scales. In terms of flood damages, both approaches have been found to produce similar results for Europe (Vousdoukas et al., 2020), which could be explained by an overestimation of the protective effect of defenses being overflown in the dynamic approach, as discussed above.

Irrespective of the type of inundation model applied, other key uncertainties relate to the accuracy of digital elevation model (discussed in the next Subsection), its resolution and, in the case of hydrodynamic approaches, data on surface roughness. As a pixel represents the average elevation height, lower resolutions lead to simplifications and smoothing of the terrain, having indications on the modeling of the flood propagation. For example, increasing the resolution of Lidar data from 100 to 10 m and using a hydrodynamic inundation model has been found to double the 100-year coastal floodplain in Faro, Portugal (Vousdoukas et al., 2018). In contrast, in North Carolina it was found that increasing the DEM resolution from 15 to 6 m using a bathtub 8-side connectivity model reduces the area below 1.1 m by factor 0.8 (Poulter & Halpin, 2008).

*Future adaptation.* Many CFR assessments do not consider human adaptation when assessing future flood risk, which leads to an overestimation (called no adaptation bias hereafter) of CFR by 2–3 orders of magnitude under SLR in 2100 (Oppenheimer et al., 2019; Hinkel et al., 2014, Table 4). There is wide consensus, both in the flood risk literature generally (Aerts et al., 2018; Di Baldassarre et al., 2015; Haer et al., 2019), and the coastal flood risks literature specifically (Hinkel et al., 2014; Oppenheimer et al., 2019; Wong et al., 2014), that assuming no adaptation is not a plausible scenario for several reasons. Coastal adaptation, specifically the form of building and enhancing coastal defenses, is widespread today, very effective in reducing CFR and societies have a long history of reducing CFR through adaptation (Charlier et al., 2005). This specifically includes these places where potentially the highest coastal flood damages could occur,



such as urban areas in river deltas, where local sea-levels have risen by up to several meters due to human-induced subsidence during the last 100 years (See Section 3.1.1). Furthermore, coastal urban areas are also those places where protection is economically very favorable, with benefits of protection being much higher than its costs during the 21st century, even under high SLR scenarios, which suggest that this approach will be widespread in the future (Aerts et al., 2014; Hallegatte et al., 2013; Hinkel et al., 2018; Lincke & Hinkel, 2018; Oppenheimer et al., 2019; Scussolini et al., 2017). As a result, the bias introduced by not considering adaptation in assessing CFR is very large.

But even when considering adaptation, uncertainties in future CFR are large, because a wide range of alternative coastal adaptation scenarios are plausible for several reasons. First, current adaptation practise is diverse, ranging from high flood hazard standards over cost-benefit analysis to large adaptation deficits (Bisaro et al., 2020; McEvoy et al., 2021; Nicholls et al., 2019). Second, adaptation has been mostly reactive depending on the experience of an extreme sea-level event (Rasmussen et al., 2021). Third, social conflicts often impede adaptation efforts and it is impossible to predict how this will evolve in the future (Hinkel et al., 2018).

Despite adaptation scenarios uncertainty being large, this has hardly been explored at broad scales. Adaptation modeling has almost exclusively focused on coastal protection and most studies have thereby focused on normative adaptation models such as maintaining protection levels (i.e., the annual probability of being flooded) constant (Hallegatte et al., 2013; Hoozemans et al., 1993; Nicholls et al., 2019; Tiggeloven et al., 2020), static cost-benefit analysis (Diaz, 2016; Nicholls et al., 2019; Tiggeloven et al., 2020; Vousdoukas et al., 2020), and robust adaptation using the criterion of benefit-cost ratios (Lincke & Hinkel, 2018). Based on these kinds of models it was found that alternative adaptation scenarios influence CFR in 2100 by factors of 20–27 (Nicholls et al., 2019, Table 4). Future work is needed to also explore other adaptation options such as accommodation, retreat and advance and it is expected that this will increase the adaptation scenarios uncertainty range. Furthermore, descriptive models, which are models aiming at mimicking actual human adaptation behavior, have not much been developed, with the exception of Hinkel et al. (2014), who applied an econometric model that explains observed protection levels through socio-economic indicators.

*Future shoreline change.* Another set of uncertainties is related to future shoreline change and the feed-back that coastal adaptation has on this in terms of preventing shoreline change. Generally, broad-scale assessments of CFR assume that the shorelines (and beach morphology) will not change, which is obviously not generally the case. Currently about 24% of the global sandy beaches are eroding at rates exceeding 0.5 m/ yr and 28% are accreting (Luijendijk et al., 2018). SLR could lead to a complete loss of 50% of the world's sandy beaches (Vousdoukas et al., 2020) and local studies show that these effects alter flood extent (Passeri et al., 2015). Furthermore, it has been shown that allowing the shoreline to retreat with SLR (rather than fixing the coastline through protection) influences tidal characteristics and extreme water levels and hence flooding (Idier et al., 2017).

#### 3.3. Exposure

*Methods.* The broad-scale assessment of exposure of land, population and assets is based on combining elevation data with spatially explicit datasets of population and land-use, and, in the case of exposure to ESL, applying inundation models for assessing the exposure to a flood with a given return period (e.g., 100-year flood). Hence, the main uncertainties in assessing current exposure are associated with the accuracy of the underlying elevation, population and asset datasets. If exposure relative to a given ESL is assessed, then uncertainties in hazard and hazard propagation as discussed in Sections 3.1 and 3.2 also play a key role. Generally, there is little information and systematic exploration of the error introduced in exposure estimates through data (in-)accuracy. However, a range of studies has explored the uncertainty in exposure through applying alternative datasets.

*Elevation data*. Large uncertainties in exposure are associated with the accuracy of near-global digital elevation models (DEM). Most of these are digital surface models (DSM) and not digital terrain models (DTM), which means they represent the elevation value of the first reflectance surface and not necessarily the terrain itself (McClean et al., 2020). The DEM widely used in earlier broad-scale coastal flood exposure and risk studies include GTOPO30, GLOBE and SRTM30, which all have a spatial resolution of 30 arc-seconds



(~1 km at the equator) (Lichter et al., 2011; McGranahan et al., 2007). More recent studies have employed the newer versions of SRTM90, which have a spatial resolution of 3 arc-seconds (approximately 90 m at the equator) or derivatives that improved SRTM such as MERIT (Yamazaki et al., 2017) and CoastalDEM (Kulp & Strauss, 2018). According to the product specification, the root mean square error (RMSE) of SRTM30 is 9.7 m (Rodriguez et al., 2005), but this has a considerable spatial variation and numerous studies have reported considerably higher accuracies, with errors of 7, 5, 2 m or even 0.5 m, particularly in coastal areas (Gorokhovich & Voustianiouk, 2006; Kellndorfer et al., 2004; Luana et al., 2015). For the Low Elevation Coastal Zone (LECZ), which is the area below 10 m that is hydrologically connected to the ocean, for example, the RMSE of SRTM90 has been estimated at 5.6 m, and those of its derivatives MERIT and CoastalDEM to 3.1 m (Gesch, 2018).

As there is a lack of global high accuracy data against which to validate global DEM, claims about the performance of global DEM in assessing flood exposure need to be taken with caution. Generally, one would expect DSM such as SRTM to underestimate flood extent. For example, the CoastalDEM correction of SRTM increases global population in the LECZ by factor 1.3 and the population in the 1-year floodplain by factor 3.7, but the neural network applied for this correction has only be trained in the US and Australia where local LIDAR (Light Detection and Ranging) data was available (Kulp & Strauss, 2018), and its performance in other regions of the world is largely unknown. Conversely, local scale analysis have found SRTM and other global DEM to overestimate both coastal (Wolff et al., 2016) and river flood extents (McClean et al., 2020). The recently released open Global Lowland DTM (GLL\_DTM\_v1), based on satellite LIDAR, has a RMSE of 0.5 m in the LECZ (Vernimmen et al., 2020), but only a horizontal resolution of 5.6 km, which makes it currently unsuitable for CFR assessment. Higher resolution versions are announced to be produced, which could then significantly improve broad-scale CFR assessments.

Further uncertainties that have not been explored at broad scales relate to the horizontal resolution of DEM (see subsection on inundation modeling above) and overlaying elevation data with the coastline and population data as these datasets don't match (Lichter et al., 2011; McGranahan et al., 2007; Neumann et al., 2015).

Current population exposure. The main uncertainties in population exposure relate to: (i) the temporal and spatial quality/accuracy of the census input population data; (ii) the implications of the methodological population redistribution approach applied; and (iii) the quality and spatial/temporal accuracies of the ancillary/covariate data used and offsets between DEMs and population data, for instance, due to different coastlines. A number of studies have explored uncertainties in global population exposure by using different global population and elevation datasets (Hinkel et al., 2014; Kulp & Strauss, 2019; Lichter et al., 2011; Mondal & Tatem, 2012; Neumann et al., 2015). Switching between the two most widely used spatial global population datasets, Global Rural-Urban Mapping Project (GRUMP) (CIESIN et al., 2011) and LandScan (Dobson et al., 2003) increases current global exposure estimates by a factor of up to 1.7, depending on the elevation data used (Table 5). This difference is due to different approaches applied and in particular the extent to which population data distribution is modeled. GRUMP is a lightly modeled population data set that assumes that people live where they are registered and that within administrative units the population is allocated to urban or rural areas, which are determined based on settlement points and night-time lights (Balk et al., 2006). Conversely LandScan is a highly modeled population data set that represents the ambient (average over 24 h) population by using roads, land cover and slope to redistribute population within administrative areas (Dobson et al., 2003). There are two more recent higher resolution global population datasets, but these have not been compared to GRUMP and LandScan. The Gridded Population of the World (GPWv4) (Center For International Earth Science Information Network-CIESIN-Columbia University, 2016) gives the non-modeled population per administrative unit and has a spatial resolution of 30 arc seconds. Worldpop (Tatem, 2017) provides a modeled population at the spatially finest resolution of 100 m. Worldpop, GPW and GRUMP have in common that they are adjusted to UN-estimates, whereas LandScan gives the ambient 24 h population and is therefore not adjusted to UN-estimates.

*Current assets exposure.* For broad-scale assessments, information about individual assets is usually not available and hence this is derived based on other datasets. One approach (Hallegatte et al., 2013; Hinkel et al., 2014) uses data on population exposure and population-to-assets multipliers that are empirically derived from global economic data such as the Penn World Table (Feenstra et al., 2015). Other approaches



(Huizinga et al., 2017) use land-use maps and national information on building construction or replacement cost density. Various datasets exist at relatively high resolution, such as the 30 m GLC30 data set (Chen et al., 2015), and products denoting the percentage of urban surface are also becoming increasingly available, such as the Global Human Settlement Layer (Pesaresi et al., 2013). These global maps represent generic urban areas not subdivided into different types of uses (e.g., commercial, industrial, etc.), which represents an important uncertainty in broad-scale assessments compared to local scale assessment, for which generally a more detailed categorization of uses is available (de Moel et al., 2015). To the best of our knowledge, no study has so far explored the uncertainty in data and methods applied for generating assets exposure at broad scales.

*Future exposure*. For the assessment of future exposure, sea-level rise, socio-economic and adaptation scenario uncertainties become relevant. Most broad-scale CFR assessments have focused on future population exposure applying national population growth to current exposure under different SSPs (Merkens et al., 2016, 2018; Neumann et al., 2015), which ignores the spatial dynamics of exposure due to processes such as urbanization, land use change and coastward migration. To address this, Jones and O' Neill (2016) and Merkens et al. (2016) created spatial explicit global population grids for each SSP based on the population projections of KC and Lutz (2017). The main difference between the approaches is that Jones and O' Neill (2016) used a gravity-based downscaling model to account for changes in urban extents, whereas Merkens et al. (2016) explicitly accounted for differences in coastal to inland population changes. These subnational projections increase exposure by up to a factor of 1.4 as compared to applying national average population growth. The effect of regionalized population projections on global population exposure has the same magnitude as the effect of switching between SSP scenarios, but a larger magnitude than switching between SLR scenarios.

#### 3.4. Vulnerability

*Methods.* Broad scale CFR assessments have focused on asset vulnerability, generally represented as depth-damage functions (DDF) that map the water depth to a relative or absolute damage. There is a wide variety of DDF, generally developed for a specific country, such as HAZUS for the USA (Scawthorn et al., 2006) or the multi-colored manual for the United Kingdom (Penning-Rowsell et al., 2014), though also a global database has been developed (Huizinga et al, 2017). Local studies apply DDF to different building types, but as such detailed data are not available at broad scales, broad scale studies apply DDF to either land-use types (Tiggeloven et al., 2020; Vousdoukas et al., 2018), or assets (Hallegatte et al., 2013; Hinkel et al., 2014). Contrary to global river flood assessments, human vulnerability, for example, in terms of flood mortality rates (Dottori et al., 2018; Jongman et al., 2015), has not been studied much in broad-scale CFR assessments.

*Depth-damage functions*. While uncertainties in DDF at broad scales have hardly been studied, a wide range of local studies has investigated the uncertainty in flood damage modeling, often focusing on river flooding, but sometimes also addressing coastal flooding. Such studies have shown that uncertainties in the damage estimation are usually the same order of magnitude or larger as uncertainties in the flood hazard assessment (de Moel & Aerts, 2011; de Moel et al., 2014; Jongman et al., 2012; Winter et al., 2018). For example, in a sensitivity analysis, de Moel et al. (2012) find that when varying flood damage model parameters, flood damages vary by up to factor 8 in breach locations on the coast of the Netherlands, with DDF being the biggest contributor to total damage uncertainty (about 30%–45%). Similar ranges have been found for coastal flooding on Small Islands (Parodi et al., 2020) and river floods in case studies in the UK and Germany (Jongman et al., 2012). On top of this, the elevation of exposed assets with respect to the ground surface can influence damage estimates considerably (Koivumäki et al., 2010). For example, flood risk reduces by 50% when assuming that the ground floor of all buildings is 50 cm above the ground surface, or 61% when assuming that all buildings would be dry flood proofed (de Moel et al., 2013).

When drawing conclusions for broad scale CFR analysis from these local findings, assumptions on heterogeneity are critically important, because when modeling a large number of individual objects, uncertainties cancel out if they are considered completely independent in terms of vulnerability and asset value (Saint-Geours et al., 2015; Wagenaar et al., 2016). Given the nature of building types in coastal cities, complete independence per object does not seem reasonable, though some heterogeneity is obviously present. Any



Overview of Uncertainties and Biases That Have Been Quantified at Global Scales

Source of uncertainty or bias	Target variable	Variation factor	Source Tables
No-adaptation bias	Flood damage	up to 1,300.0	4
Adaptation scenario uncertainty	Flood damage	20.0-26.7	4
Socio-economic scenarios uncertainty	Flood exposure, flood damage	2.3-5.8	4
Ice sheet model uncertainty	Mean sea-level	1.6-3.8	1
Digital elevation model uncertainty	Flood extent, area exposure, people exposure	1.2-3.8	5
Emission scenario uncertainty (mean SLR forcing)	Mean sea-level, extreme sea-levels, flood damage	1.6-2.1	1,4
Population data uncertainty	Population exposure	1.1–1.7	5
Wave calibration datasets uncertainty	Significant wave height	1.7	3
Inundation model bias/uncertainty*	Flood damages	1.6	4
EVA method uncertainty (waves and surges)	Extreme sea-levels, significant wave height	1.1–1.5	2,3
Regionalization of population projections uncertainty	Flood exposure	1.4	5
Emission scenario uncertainty (atmospheric forcing)	Extreme sea-levels	0.8-1.1	2
No-wave-set-up bias	Extreme sea-levels	1.1	5

\*The study from which the number is taken covers only Europe.

bias in the functional form of the DDF chosen, however, will not be canceled out. Overall, studies show that between sets of damage functions there are large differences, but uncertainty estimates within a set of functions, generally associated with the value at risk, is substantially lower, as is illustrated by the different estimates of the Multi-Colored Manual (Penning-Rowsell, 2013) and the global database developed by the JRC (Huizinga et al., 2017). Further uncertainties, which have however not been explored at broad scales, relate to water depth being the only variable taken into account in the calculation of damage with DDF, which means that other factors also mediating the damage caused, such as hydraulic pressure, wave forces and salinity, are not considered.

*Future vulnerability*. Projecting future CFR requires understanding the vulnerability of future societies. While globally, both human and economic vulnerability has decreased significantly in recent decades (Bouwer & Jonkman, 2018; Formetta & Feyen, 2019; Kreibich et al., 2017), the future dynamics in vulnerability (e.g., due to improved early warning and emergency response, flood proofing of infrastructures, better health care, more resistant building practices etc.), have not been addressed in broad-scale CFR assessments studies.

# 4. Discussion

#### 4.1. What are the Major Uncertainties and Biases and How Could They be Reduced?

When comparing uncertainties and biases across dimensions of broad-scale CFR assessments, by far the largest ones are associated with future human adaptation behavior, which potentially has a multiple order of magnitude effect on future flood risk (Table 7). The largest part of this effect is due to the no-adaptation bias, resulting from ignoring coastal adaptation, which is still a default assumption in many broad-scale CFR assessments (and which dominates associated media coverage). This assumption is, however, mislead-ing and should not be included in the sets of adaptation scenarios that inform policy. In today's world we see extensive adaptation (mainly protection) in coastal areas, high economic benefits thereof and widespread discussion and plans for further adaptation. Hence, it can be asserted that no adaptation scenarios describe a future that will never be seen. The other part of this effect is adaptation scenario uncertainty (factors 20–27), which cannot, by definition, be reduced in principle, because it is not possible to predict how societies will adapt in the future (Oppenheimer et al., 2019). A diversity of adaptation behaviors is plausible, ranging from hard engineered to nature-based solutions and coastal retreat, which need to be explored in broad scale CFR assessments. This can be supported by developing sets of plausible adaptation scenarios that can be used consistently across modeling efforts, similarly to the SSPs.

Next down the ladder of highest global uncertainties are those associated with socio-economic development, ice sheet models and digital elevation data influencing CFR by factors of 2.3–5.8, 1.6–3.8 and 1.2–3.8, respectively (Table 7). Regarding the digital elevation data, it has been argued that a collaborative and open effort toward a freely available high accuracy DEM is needed to address this limitation (Gesch, 2018; Mc-Clean et al., 2020; Schumann & Bates, 2018). A higher resolution version of the newly released and freely available satellite LIDAR-based Global Lowland digital terrain model by Vernimmen et al. (2020) may constitute a big step in this direction.

This is followed by GHG emission uncertainty (mean SLR forcing), which contributes with factors of up 1.6 to 2.1 to global mean sea-level uncertainty. Uncertainties in population data, wave model calibration datasets and inundation models have not been studied much at broad scales, but in the few available broad-scale studies these are globally relevant with factors of around 1.5. Emission uncertainty due to atmospheric forcing driving changes in surges and waves is comparably small. The same holds true for the bias introduced by not considering the wave set up contribution globally. While being small in terms of global mean SLR uncertainty, the contribution of human-induced subsidence to local relative sea-level rise in river deltas, especially the rapid subsidence observed in many urban areas in such settings, is globally substantial at least until 2050. Given that this is influenced directly by human agency, however, future human-induced subsidence is highly uncertain.

There are also uncertainties that have been observed to be significant locally, but that have not been quantified at broad scales. Concerning the sea-level hazard, by far the largest uncertainty is related to maximum surge and wave heights under tropical cyclones (Tables 2 and 3). For example, locally surge models have underestimated ESL of large return periods during tropical cyclones by up to a factor of 4 (Muis et al., 2016). What this means for flood risk, which integrates over all return periods, is yet unknown, but as a large fraction of the global coastal zone is exposed to tropical cyclones, these uncertainties are expected to also be significant at global scale. Other uncertainties in sea-level hazard are locally significant for some regions, but probably not so relevant globally. This includes uncertainties due to local shallow water interactions between SLR, tides, waves and surges. In locations with shallow water slopes or extensive tidal flats, such as the German Bight, these processes can double ESL in 2100 (Arns et al., 2017). Ongoing efforts to implement numerical models fed with all forcings is expected to reduce these uncertainties.

Concerning the other components of CFR, uncertainties that have been observed to be significant locally, but that have not been quantified at broad scales, include, foremost, uncertainties in depth-damage functions, for which local studies have shown an effect on flood damages by up to factor 16 (Table 6). Related to this, uncertainties in asset exposure are also expected to be significant globally, given the prevalent large uncertainties in estimates of local GDP and fixed capital stock (Huizinga et al., 2017). With regards to hazard propagation, large uncertainties lie in levels, quality and associated failure modes of coastal defenses. The latter, for example, has been shown to affect flood damages by a factor of up to 4 (Wadey et al., 2012), which is much larger than the local biases reported in association to the bathtub approach (factors 0.5 to 2). Addressing these uncertainties requires the collection of large amounts of local data on observed depth-damage relationships and defense failures, and this would, similarly to what was described above for digital elevation data, also benefit from open community efforts.

## 4.2. What do These Uncertainties Mean for the Interpretation of CFR Assessment?

When interpreting our results, it is important to recognize that the information on global uncertainties available in the literature is limited, and as a result, we may have overestimated or underestimated some uncertainties. For some dimensions of uncertainties such as defense levels, defense failure and depth damage models, for example, only local estimates were available. Local estimates are generally higher than broad-scale estimates, because in aggregation some of the local uncertainties are canceled out. Furthermore, local studies are generally conducted where uncertainties are expected to be particularly large, which can distort the broad-scale picture. But uncertainties may also be overestimated in broad-scale studies themselves, because many studies have only explored uncertainty ranges, ignoring that the likelihood of the "true" value is generally not uniformly distributed across these ranges.

Another limitation inherent in our literature-based approach to CFR uncertainty as compared to a full *global sensitivity analysis* (see Section 2) is that we were not able to vary all uncertain variables simultaneously and hence may not have captured all interactions between uncertain variables. It turns out, however, that the most important interactions, specifically between different components of CFR, have been covered by the available studies. Hence, it is highly unlikely that conducting a global sensitivity analysis would change the results of this paper significantly. Arguably the major source of nonlinearity in CFR assessments comes from the non-linear distribution of exposure across elevation levels. This could lead to the situation that under low SLR (and hence lower ESL) variations in exposure (or vulnerability, adaptation, etc.) may only have a small effect on CFR, while under high SLR the effect would be large. Hence, it is important to assess interactions specifically between SLR and other uncertain variables, which is generally done by CFR studies because SLR is the main motivation of these studies in the first place. For example, the high sensitivity of damages to adaptation is found for the full range of plausible 21st century sea-level rise scenarios (Table 4).

Nevertheless, uncertainties in broad-scale CFR assessments remain substantial and their results must be understood as indicative of the broad magnitude of flood risks. For the purpose of addressing some of the broad-scale economic questions listed in the Introduction (e.g., global cost of adaptation, risk transfer mechanisms, disaster relief funds, etc.), this is generally not problematic because outputs of CFR, such as expected annual flood damages, must be seen in relation to other economic variables such GDP, population or asset density, which all differ by several orders of magnitude between regions and countries. Furthermore, annual GDP growth rates and discount rates, which typically lie in the order of several percent, inflate or deflate economic variables faster on decadal time scales than sea-levels rise. Both of these points are, for example, illustrated in broad scale cost-benefit analysis of coastal protection measures (Diaz, 2016; Hinkel et al., 2018; Tiggeloven et al., 2020; Vousdoukas et al., 2020), which show order of magnitude differences between benefit-cost ratios around the world's coasts. The same holds true for expected annual coastal flood damages relative to national GDP (Hinkel et al., 2012; Lincke & Hinkel, 2018; Tol, 2007).

Against this background, a number of simplifications made in broad-scale CFR assessments can be justified or are even necessary in order to enable the large number of simulations necessary for some economic analysis. One example is the use of the bathtub model in economic optimization approaches, because the large number of simulations required for this can generally not be conducted with hydrodynamic flood models (Lincke & Hinkel, 2018; Tiggeloven et al., 2020). At the same time, there is a need to better explore the biases in both bathtub and reduced-complexity hydrodynamic models in conjunction with defense failure modes, which has hardly been investigated at broader scales. Another example of a feasible simplification is to ignore changes in ESL distribution due to changing wind and pressure fields under different emission scenarios (atmospheric forcing) and climate models (also called changes in storminess). Given that the contribution of these processes to changes in future ESL is about one order of magnitude smaller than the contribution of changes in mean sea-levels to future ESL (Table 7), substantial computation time can be saved by displacing extreme water levels upwards with mean sea-levels only, instead of also running tide-surge models forced with wind and pressure fields from climate models.

For other purposes such as flood hazard mapping, or when zooming into a particular location, epistemic uncertainties in mean and extreme sea-level hazards, though smaller than socio-economic ones, become very relevant. Determining how safe coastal populations are in a given place requires a much higher accuracy in the water levels and elevation than the broad scale, more economically oriented studies.

Finally, we note that we only considered CFR in terms of direct damages, ignoring the propagation of these through the economic and financial systems to cause indirect economic damages, which extend beyond the flooded area. For example, floods interrupt the production and flow of goods, decrease labor productivity of those affected, and land permanently submerged leads to a loss of land input into agricultural production. This, together with rising adaptation costs, increases the demand for construction services and capital, and hence public debt (Parrado et al., 2020), which in turn propagates through global economic (Bosello et al., 2012; Parrado et al., 2020; Schinko et al., 2020) and financial (Mandel et al., 2020) networks. Locally, and in the direct aftermath of disasters, indirect effects increase overall damages. For example, this has been estimated to account for 40% of the total damages in the case of Hurricane Katrina (Hallegatte, 2008). Longer term macroeconomic effects of disasters can be both positive and negative (Lazzaroni & van Bergeijk, 2014), and one major uncertainty thereby is the economic dynamics of recovery after the event (Koks



et al., 2019). We do not consider these effects here, because their assessment relies on wider assumptions on the whole economy, which cannot be covered in this study.

# 5. Conclusions

This study provided a literature-based comparative assessment of the major sources of uncertainty and bias in broad-scale CFR assessment, considering the four components of: (i) mean and extreme sea-level hazards (including sea-level rise, tides, surges, waves, river run-off and their interactions); (ii) propagation of these hazards into the floodplain, including their interaction with natural and artificial flood barriers; (iii) exposure in terms of area, people and coastal assets threatened by these hazards; and (iv) vulnerability, mainly in terms of depth-damage functions of coastal assets.

We find that there are substantial uncertainties in all dimensions, which highlights the interdisciplinary nature of CFR assessments and the need for dedicated disciplinary efforts to improve the assessment of each dimension. At the same time, there is the need to collectively work across disciplines and toward the needs of the global policy community, as the importance of some issues/uncertainties only become apparent when looking beyond disciplinary bounds. For example, while from a disciplinary perspective there are many uncertainties worth exploring in sea level, from a transdisciplinary perspective of supporting global policy, uncertainties in the order of 10 or 20 cm higher or lower sea-levels may be secondary to many of the other uncertainties raised in this paper.

Globally, by far the largest bias in assessing future CFR is introduced by not considering human adaptation, which can lead to an overestimation of CFR in 2100 by up to factor 1300. But even when considering adaptation, uncertainties in how coastal societies will adapt to sea-level rise dominate, with a factor of up to 27, all other uncertainties. Other large uncertainties that have been quantified globally are associated with socio-economic development (factors 2.3–5.8), digital elevation data (factors 1.2–3.8), ice sheet models (factor 1.6–3.8) and greenhouse gas emissions (factors 1.6–2.1). Local uncertainties that stand out but have not been quantified globally, relate to depth-damage functions, defense failure mechanisms, surge and wave heights in areas affected by tropical cyclones (in particular for large return periods), as well as nearshore interactions between mean sea-levels, storm surges, tides and waves.

We conclude that the advancement of broad-scale CFR assessment requires more comprehensive analysis of uncertainties, including considering uncertainties in methods and data, which have received little attention so far. In particular, adaptation should be considered more explicitly and community efforts to develop consistent adaptation scenarios to be tested in CFR models would be beneficial. The reduction of epistemic uncertainties in hazards requires continued incorporation of new developments in numerical and statistical modeling, specifically taking into account the non-linear interactions between mean sea-level rise, surges, tides and waves. Finally, reducing epistemic uncertainties in digital elevation, coastal protection levels and depth-damage functions requires open community-based efforts, in which many scholars work together in quantifying and validating data from multiple sources.

# **Data Availability Statement**

Data were not used, nor created for this research.

# References

- Aarnes, O. J., Reistad, M., Breivik, Ø., Bitner-Gregersen, E., Eide, L. I., Gramstad, O., et al. (2017). Projected changes in significant wave height toward the end of the 21st century: Northeast Atlantic. Journal of Geophysical Research: Oceans, 122, 3394–3403. https://doi. org/10.1002/2016JC012521
- Abadie, L. M. (2018). Sea level damage risk with probabilistic weighting of IPCC scenarios: An application to major coastal cities. Journal of Cleaner Production, 175, 582–598. https://doi.org/10.1016/j.jclepro.2017.11.069
- Abadie, L. M., Sainz de Murieta, E., & Galarraga, I. (2016). Climate risk assessment under uncertainty: An application to main European coastal cities. *Frontiers in Marine Science*, 3. https://doi.org/10.3389/fmars.2016.00265
- Aerts, J. C. J. H., Botzen, W. J., Clarke, K. C., Cutter, S. L., Hall, J. W., Merz, B., et al. (2018). Integrating human behaviour dynamics into flood disaster risk assessment. *Nature Climate Change*, *8*, 193–199. https://doi.org/10.1038/s41558-018-0085-1

Aerts, J. C. J. H., Botzen, W. J. W., Emanuel, K., Lin, N., de Moel, H., & Michel-Kerjan, E. O. (2014). Evaluating flood resilience strategies for coastal megacities. *Science*, 344, 473–475. https://doi.org/10.1126/science.1248222

#### Acknowledgments

The authors thank all participants of the COASTMIP project (www.coastmip. org) for the excellent discussions we had at our meetings, which supported the thinking that went into this paper. The authors also thank two anonymous reviewers for their very helpful comments. DL, GLC, JH, MM, RvdW have been partially supported by the ERA4CS Project INSeaPTION (grant 01LS1703A) and DL and JH by the Project ISIPEDIA (grant 01LS1711C). Both Projects are part of ERA4CS, an ERA-NET initiated by JPI Climate and funded by FORMAS (SE), BMBF (DE), BMWFW (AT), IFD (DK), MINECO (ES), ANR (FR) with co-funding by the European Union (Grant 690462). DL, JH, RN and RvdW have received funding from the PROTECT project (grant 869304), and DL and JH from the COACCH project (grant 776479), both funded under the European Union's Horizon 2020 Research and Innovation Programme. ATV, CW, DL and JH have been supported by the Deutsche Forschungsgemeinschaft (DFG) through the SEASCAPE II project, which is part of the Special Priority Program 1889 Regional Sea Level Change and Society. MM has been partially supported by the MOCCA project (grant RTI2018-093941-B-C31). TW was partially supported by the National Aeronautics and Space Administration (NASA) under the New (Early Career) Investigator Program in Earth Science (grant number: 80NSSC18K0743) and MH was partially supported by the Australian Government National Environmental Science Program Earth Systems and Climate Change Hub. PJW has been supported by the Dutch Research Council (NWO) in the form of a VIDI grant (grant no. 016.161.324).



- Allsop, W., Kortenhaus, A., & Morris, M. (2007). Failure mechanisms for flood defense structures (No. T04-06–01), FLOODsite project report. (Contract No:GOCE-CT-2004-505420. HR Wallingford Ltd.
- Amores, A., Marcos, M., Carrió, D. S., & Gómez-Pujol, L., (2020). Coastal impacts of Storm Gloria (January 2020) over the north-western Mediterranean. Natural Hazards and Earth System Sciences 20, 1955–1968. https://doi.org/10.5194/nhess-20-1955-2020
- Appendini, C. M., Pedrozo-Acuña, A., Meza-Padilla, R., Torres-Freyermuth, A., Cerezo-Mota, R., López-González, J., & Ruiz-Salcines, P. (2017). On the role of climate change on wind waves generated by tropical cyclones in the Gulf of Mexico. *Coastal Engineering Journal*, 59, 1740001. https://doi.org/10.1142/S0578563417400010
- Arns, A., Dangendorf, S., Jensen, J., Talke, S., Bender, J., & Pattiaratchi, C. (2017). Sea-level rise induced amplification of coastal protection design heights. *Scientific Reports*, 7, 40171. https://doi.org/10.1038/srep40171
- Arns, A., Wahl, T., Dangendorf, S., & Jensen, J. (2015). The impact of sea level rise on storm surge water levels in the northern part of the German Bight. *Coastal Engineering*, *96*, 118–131. https://doi.org/10.1016/j.coastaleng.2014.12.002
- Arns, A., Wahl, T., Wolff, C., Vafeidis, A. T., Haigh, I. D., Woodworth, P., et al., (2020). Non-linear interaction modulates global extreme sea levels, coastal flood exposure, and impacts. *Nature Communications* 11, 1–9. https://doi.org/10.1038/s41467-020-15752-5
- Athanasiou, P., van Dongeren, A., Giardino, A., Vousdoukas, M., Gaytan-Aguilar, S., & Ranasinghe, R. (2019). Global distribution of nearshore slopes with implications for coastal retreat. *Earth System Science Data*, 11, 1515–1529. https://doi.org/10.5194/essd-11-1515-2019
- Balk, D. L., Deichmann, U., Yetman, G., Pozzi, F., Hay, S. I., & Nelson, A. (2006). Determining global population distribution: Methods, applications and data. In Hay, S. I., Graham, A., & Rogers, D. J. (Eds.), Advances in parasitology, global mapping of infectious diseases: Methods, examples and emerging applications (pp. 119–156). Academic Press. https://doi.org/10.1016/S0065-308X(05)62004-0
- Bamber, J. L., Oppenheimer, M., Kopp, R. E., Aspinall, W. P., & Cooke, R. M. (2019). Ice sheet contributions to future sea-level rise from structured expert judgment. Proceedings of the National Academy of Sciences of the United States of America, 116, 11195–11200. https:// doi.org/10.1073/pnas.1817205116
- Bates, P. D., Horritt, M. S., & Fewtrell, T. J. (2010). A simple inertial formulation of the shallow water equations for efficient two-dimensional flood inundation modelling. *Journal of Hydrology*, 387, 33–45. https://doi.org/10.1016/j.jhydrol.2010.03.027
- Beck, M. W., Losada, I. J., Menéndez, P., Reguero, B. G., Díaz-Simal, P., & Fernández, F. (2018). The global flood protection savings provided by coral reefs. *Nature Communications*, 9, 2186. https://doi.org/10.1038/s41467-018-04568-z
- Beetham, E., & Kench, P. S. (2018). Predicting wave overtopping thresholds on coral reef-island shorelines with future sea-level rise. Nature Communications, 9, 3997. https://doi.org/10.1038/s41467-018-06550-1
- Bertin, X., Li, K., Roland, A., & Bidlot, J.-R. (2015). The contribution of short-waves in storm surges: Two case studies in the Bay of Biscay. Continental Shelf Research, 96, 1–15. https://doi.org/10.1016/j.csr.2015.01.005
- Bisaro, A., de Bel, M., Hinkel, J., Kok, S., Stojanovic, T., & Ware, D. (2020). Multilevel governance of coastal flood risk reduction: A public finance perspective. *Environmental Science & Policy*, 112, 203–212. https://doi.org/10.1016/j.envsci.2020.05.018
- Bloemendaal, N., Haigh, I. D., de Moel, H., Muis, S., Haarsma, R. J., & Aerts, J. C. J. H. (2020). Generation of a global synthetic tropical cyclone hazard dataset using STORM. Scientific Data, 7, 40. https://doi.org/10.1038/s41597-020-0381-2
- Bloemendaal, N., Muis, S., Haarsma, R. J., Verlaan, M., Irazoqui Apecechea, M., de Moel, H., et al. (2018). Global modeling of tropical cyclone storm surges using high-resolution forecasts. *Climate Dynamics*, 52, 5031–5044. https://doi.org/10.1007/s00382-018-4430-x
- Booij, N., Holthuijsen, L. H., & Ris, R. C. (1997), The "Swan" wave model for shallow water. In Coastal engineering 1996. (pp. 668–676). https://doi.org/10.1061/9780784402429.053
- Bosello, F., Nicholls, R. J., Richards, J., Roson, R., & Tol, R. S. J. (2012). Economic impacts of climate change in Europe: Sea-level rise. *Climatic Change*, 112, 63–81. https://doi.org/10.1007/s10584-011-0340-1
- Bouwer, L. M., & Jonkman, S. N. (2018). Global mortality from storm surges is decreasing. *Environmental Research Letters*, 13, 014008. https://doi.org/10.1088/1748-9326/aa98a3
- Breilh, J. F., Chaumillon, E., Bertin, X., & Gravelle, M. (2013). Assessment of static flood modeling techniques: Application to contrasting marshes flooded during Xynthia (western France). Natural Hazards and Earth System Sciences, 13, 1595–1612. https://doi.org/10.5194/ nhess-13-1595-2013
- Brown, S., Nicholls, R. J., Lowe, J. A., & Hinkel, J. (2016). Spatial variations of sea-level rise and impacts: An application of DIVA. *Climatic Change*, 134, 403–416. https://doi.org/10.1007/s10584-013-0925-y
- Brown, S., Nicholls, R. J., Pardaens, A. K., Lowe, J. A., Tol, R. S. J., Vafeidis, A. T., & Hinkel, J. (2019). Benefits of climate-change mitigation for reducing the impacts of sea-level rise in G-20 countries. *Journal of Coastal Research*, 35, 884. https://doi.org/10.2112/ JCOASTRES-D-16-00185.1
- Buchanan, M. K., Oppenheimer, M., & Kopp, R. E. (2017). Amplification of flood frequencies with local sea level rise and emerging flood regimes. Environmental Research Letters, 12, 064009. https://doi.org/10.1088/1748-9326/aa6cb3
- Calafat, F. M., Avgoustoglou, E., Jordà, G., Flocas, H., Zodiatis, G., Tsimplis, M. N., & Kouroutzoglou, J. (2014). The ability of a barotropic model to simulate sea level extremes of meteorological origin in the Mediterranean Sea, including those caused by explosive cyclones. *Journal of Geophysical Research: Oceans*, 119, 7840–7853. https://doi.org/10.1002/2014JC010360
- Calafat, F. M., & Marcos, M. (2020). Probabilistic reanalysis of storm surge extremes in Europe. Proceedings of the National Academy of Sciences, 117, 1877–1883. https://doi.org/10.1073/pnas.1913049117
- Camus, P., Cofiño, A. S., Mendez, F. J., & Medina, R. (2011). Multivariate wave climate using self-organizing maps. Journal of Atmospheric and Oceanic Technology, 28, 1554–1568. https://doi.org/10.1175/JTECH-D-11-00027.1
- Center For International Earth Science Information Network-CIESIN-Columbia University. (2016). Gridded population of the world, Version 4(GPWv4): Population count. https://doi.org/10.7927/H4X63JVC
- Charlier, R. H., Chaineux, M. C. P., & Morcos, S. (2005). Panorama of the history of coastal protection. *Journal of Coastal Research*, 211, 79–111. https://doi.org/10.2112/03561.1
- Chen, J., Chen, J., Liao, A., Cao, X., Chen, L., Chen, X., et al. (2015). Global land cover mapping at 30m resolution: A POK-based operational approach. *ISPRS Journal of Photogrammetry and Remote Sensing, Global Land Cover Mapping and Monitoring, 103, 7–27.* https:// doi.org/10.1016/j.isprsjprs.2014.09.002
- Cheon, S.-H., & Suh, K.-D. (2016). Effect of sea level rise on nearshore significant waves and coastal structures. *Ocean Engineering*, 114, 280–289. https://doi.org/10.1016/j.oceaneng.2016.01.026
- Christensen, P., Gillingham, K., & Nordhaus, W. (2018). Uncertainty in forecasts of long-run economic growth. *Proceedings of the National Academy of Sciences*, 115, 5409–5414. https://doi.org/10.1073/pnas.1713628115
- Cid, A., Wahl, T., Chambers, D. P., & Muis, S. (2018). Storm surge reconstruction and return water level estimation in Southeast Asia for the 20th Century. *Journal of Geophysical Research: Oceans*, 123, 437–451. https://doi.org/10.1002/2017JC013143



- CIESIN, IFPRI, The World Bank, & CIAT. (2011). Global rural-urban mapping project, version 1 (GRUMPv1): Population density grid. Data and Applications Center (SEDAC), Columbia University.
- Collins, M., Knutti, R., Arblaster, J., Dufresne, J.-L., Fichefet, T., Friedlingstein, P., et al. (2013). Long-term climate change: Projections, commitments and irreversibility. In Climate change 2013-The physical science basis: Contribution of working group I to the fifth assessment report of the intergovernmental panel on climate change (pp. 1029–1136). https://doi.org/10.1017/CB09781107415324.024
- Conte, D., & Lionello, P. (2013). Characteristics of large positive and negative surges in the Mediterranean Sea and their attenuation in future climate scenarios. *Global and Planetary Change*, 111, 159–173. https://doi.org/10.1016/j.gloplacha.2013.09.006
- Cowell, P. J., Roy, P. S., & Jones, R. A. (1995). Simulation of large-scale coastal change using a morphological behaviour model. Marine Geology, 126, 45–61. https://doi.org/10.1016/0025-3227(95)00065-7
- DeConto, R. M., & Pollard, D. (2016). Contribution of Antarctica to past and future sea-level rise. *Nature*, 531, 591–597. https://doi. org/10.1038/nature17145
- de Moel, H., & Aerts, J. C. J. H. (2011). Effect of uncertainty in land use, damage models and inundation depth on flood damage estimates. Natural Hazards, 58, 407–425. https://doi.org/10.1007/s11069-010-9675-6
- de Moel, H., Asselman, N. E. M., & Aerts, J. C. J. H. (2012). Uncertainty and sensitivity analysis of coastal flood damage estimates in the west of the Netherlands. Natural Hazards and Earth System Sciences, 12, 1045–1058. https://doi.org/10.5194/nhess-12-1045-2012
- de Moel, H., Bouwer, L. M., & Aerts, J. C. J. H. (2014). Uncertainty and sensitivity of flood risk calculations for a dike ring in the south of the Netherlands. *Science of the Total Environment*, 473–474, 224–234. https://doi.org/10.1016/j.scitotenv.2013.12.015
- de Moel, H., Jongman, B., Kreibich, H., Merz, B., Penning-Rowsell, E., & Ward, P. J. (2015). Flood risk assessments at different spatial scales. *Mitigation and Adaptation Strategies for Global Change*, 20, 865–890. https://doi.org/10.1007/s11027-015-9654-z
- de Moel, H., van Vliet, M., & Aerts, J. C. J. H. (2013). Evaluating the effect of flood damage-reducing measures: A case study of the unembanked area of Rotterdam, the Netherlands. *Regional Environmental Change*. https://doi.org/10.1007/s10113-013-0420-z
- Diaz, D. B. (2016). Estimating global damages from sea level rise with the Coastal Impact and Adaptation Model (CIAM). *Climatic Change*, 137, 143–156. https://doi.org/10.1007/s10584-016-1675-4
- Di Baldassarre, G., Viglione, A., Carr, G., Kuil, L., Yan, K., Brandimarte, L., & Blöschl, G. (2015). Debates—Perspectives on socio-hydrology: Capturing feedbacks between physical and social processes. *Water Resources Research*, 51, 4770–4781. https://doi. org/10.1002/2014WR016416
- Dobson, J. E., Brlght, E. A., Coleman, P. R., Durfee, R. C., & Worley, B. A. (2003). LandScan: A global population database for estimating populations at risk. *Photogrammetric Engineering & Remote Sensing*, 66, 849–314. https://doi.org/10.1201/9781482264678-24
- Dodet, G., Melet, A., Ardhuin, F., Bertin, X., Idier, D., & Almar, R. (2019). The contribution of wind-generated waves to coastal sea-level changes. Surveys in Geophysics, 40, 1563–1601. https://doi.org/10.1007/s10712-019-09557-5
- Dottori, F., Szewczyk, W., Ciscar, J.-C., Zhao, F., Alfieri, L., Hirabayashi, Y., et al. (2018). Increased human and economic losses from river flooding with anthropogenic warming. *Nature Climate Change*, 8, 781–786. https://doi.org/10.1038/s41558-018-0257-z
- Du, J., Shen, J., Zhang, Y. J., Ye, F., Liu, Z., Wang, Z., et al. (2018). Tidal Response to sea-level rise in different types of estuaries: The importance of length, bathymetry, and geometry: Tidal response to sea-level rise. *Geophysical Research Letters*, 45, 227–235. https://doi.org/10.1002/2017GL075963
- Eilander, D., Couasnon, A., Ikeuchi, H., Muis, S., Yamazaki, D., Winsemius, H., & Ward, P. J. (2020). The effect of surge on riverine flood hazard and impact in deltas globally. *Environmental Research Letters*, *15*, 104007. https://doi.org/10.1088/1748-9326/ab8ca6
- Emanuel, K., Ravela, S., Vivant, E., & Risi, C. (2006). A statistical deterministic approach to hurricane risk assessment. Bulletin of the American Meteorological Society, 87, 299–314. https://doi.org/10.1175/BAMS-87-3-299
- Erkens, G., Bucx, T., Dam, R., de Lange, G., & Lambert, J., (2015). Sinking coastal cities. In Proceedings of the international association of hydrological sciences. Presented at the prevention and mitigation of natural and anthropogenic hazards due to land subsidence–Ninth international symposium on land subsidence (NISOLS). (pp. 189–198). Nagoya, Japan: Copernicus GmbH. https://doi.org/10.5194/ piahs-372-189-2015
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9, 1937–1958. https://doi. org/10.5194/gmd-9-1937-2016
- Feenstra, R. C., Inklaar, R., & Timmer, M. P. (2015). The next generation of the Penn World table. American Economic Review, 105, 3150– 3182. https://doi.org/10.1257/aer.20130954
- Formetta, G., & Feyen, L. (2019). Empirical evidence of declining global vulnerability to climate-related hazards. *Global Environmental Change*, 57, 101920. https://doi.org/10.1016/j.gloenvcha.2019.05.004
- Frieler, K., Lange, S., Piontek, F., Reyer, C. P. O., Schewe, J., Warszawski, L., et al. (2017). Assessing the impacts of 1.5°C global warming – Simulation protocol of the Inter-Sectoral Impact Model Intercomparison Project (ISIMIP2b). *Geoscientific Model Development*, 10, 4321–4345. https://doi.org/10.5194/gmd-10-4321-2017
- Gallien, T. W. (2016). Validated coastal flood modeling at Imperial Beach, California: Comparing total water level, empirical and numerical overtopping methodologies. *Coastal Engineering*, 111, 95–104. https://doi.org/10.1016/j.coastaleng.2016.01.014
- Gemmrich, J., Thomas, B., & Bouchard, R. (2011). Observational changes and trends in northeast Pacific wave records. *Geophysical Research Letters*, 38, L22601. https://doi.org/10.1029/2011GL049518
- Gesch, D. B. (2018). Best Practices for elevation-based assessments of sea-level rise and coastal flooding exposure. Frontiers of Earth Science, 6. https://doi.org/10.3389/feart.2018.00230
- Goodwin, I. D., Stables, M. A., & Olley, J. M. (2006). Wave climate, sand budget and shoreline alignment evolution of the Iluka-Woody Bay sand barrier, northern New South Wales, Australia, since 3000 yr BP. *Marine Geology*, 226, 127–144. https://doi.org/10.1016/j. margeo.2005.09.013
- Gorokhovich, Y., & Voustianiouk, A. (2006). Accuracy assessment of the processed SRTM-based elevation data by CGIAR using field data from USA and Thailand and its relation to the terrain characteristics. *Remote Sensing of Environment, 104*, 409–415. https://doi.org/10.1016/j.rse.2006.05.012
- Haer, T., Botzen, W. J. W., & Aerts, J. C. J. H. (2019). Advancing disaster policies by integrating dynamic adaptive behaviour in risk assessments using an agent-based modelling approach. *Environmental Research Letters*, 14, 044022. https://doi.org/10.1088/1748-9326/ ab0770
- Haigh, I. D., MacPherson, L. R., Mason, M. S., Wijeratne, E. M. S., Pattiaratchi, C. B., Crompton, R. P., & George, S. (2014). Estimating present day extreme water level exceedance probabilities around the coastline of Australia: Tropical cyclone-induced storm surges. *Climate Dynamics*, 42, 139–157. https://doi.org/10.1007/s00382-012-1653-0



- Haigh, I. D., Pickering, M. D., Green, J. A. M., Arbic, B. K., Arns, A., Dangendorf, S., et al. (2020). The Tides They Are a-Changin': A comprehensive review of past and future non-astronomical changes in tides, their driving mechanisms, and future implications. *Reviews of Geophysics*, 58. https://doi.org/10.1029/2018RG000636
- Hall, J. W., Dawson, R. J., Sayers, P. B., Rosu, C., Chatterton, J. B., & Deakin, R. (2003). A methodology for national-scale flood risk assessment. Proceedings of the Institution of Civil Engineers–Water and Maritime Engineering, 156, 235–247. https://doi.org/10.1680/ wame.2003.156.3.235
- Hallegatte, S. (2008). An adaptive regional input-output model and its application to the assessment of the economic cost of Katrina. *Risk Analysis*, 28, 779–799. https://doi.org/10.1111/j.1539-6924.2008.01046.x
- Hallegatte, S., Green, C., Nicholls, R. J., & Corfee-Morlot, J. (2013). Future flood losses in major coastal cities. *Nature Climate Change*, *3*, 802–806. https://doi.org/10.1038/nclimate1979
- Hamdi, Y., Bardet, L., Duluc, C.-M., & Rebour, V. (2014). Extreme storm surges: A comparative study of frequency analysis approaches. Natural Hazards and Earth System Sciences, 14, 2053–2067. https://doi.org/10.5194/nhess-14-2053-2014
- Hanson, S., Nicholls, R., Ranger, N., Hallegatte, S., Corfee-Morlot, J., Herweijer, C., & Chateau, J. (2011). A global ranking of port cities with high exposure to climate extremes. *Climatic Change*, 104, 89–111. https://doi.org/10.1007/s10584-010-9977-4
- Hausfather, Z., & Peters, G. P. (2020). Emissions The 'business as usual' story is misleading. Nature, 577, 618–620. https://doi.org/10.1038/ d41586-020-00177-3
- Herrera-García, G., Ezquerro, P., Tomás, R., Béjar-Pizarro, M., López-Vinielles, J., Rossi, M., et al. (2021). Mapping the global threat of land subsidence. *Science*, 371, 34–36. https://doi.org/10.1126/science.abb8549
- Hinkel, J., Aerts, J. C. J. H., Brown, S., Jiménez, J. A., Lincke, D., Nicholls, R. J., et al. (2018). The ability of societies to adapt to twenty-first-century sea-level rise. *Nature Climate Change*, *8*, 570–578. https://doi.org/10.1038/s41558-018-0176-z
- Hinkel, J., Brown, S., Exner, L., Nicholls, R. J., Vafeidis, A. T., & Kebede, A. S. (2012). Sea-level rise impacts on Africa and the effects of mitigation and adaptation: An application of DIVA. *Regional Environmental Change*, *12*, 207–224. https://doi.org/10.1007/s10113-011-0249-2
- Hinkel, J., Lincke, D., Vafeidis, A. T., Perrette, M., Nicholls, R. J., Tol, R. S. J., et al. (2014). Coastal flood damage and adaptation costs under 21st century sea-level rise. *Proceedings of the National Academy of Sciences*, 111, 3292–3297. https://doi.org/10.1073/pnas.1222469111
   Hodges, K., Cobb, A., & Vidale, P. L. (2017). How well are tropical cyclones represented in reanalysis datasets? *Journal of Climate*, 30,
- 5243-5264. https://doi.org/10.1175/JCLI-D-16-0557.1
- Holthuijsen, L. H. (2010). Waves in oceanic and coastal waters. Cambridge: Cambridge University Press. https://doi.org/10.1017/ CBO9780511618536
- Hongo, C., Kurihara, H., & Golbuu, Y. (2018). Projecting of wave height and water level on reef-lined coasts due to intensified tropical cyclones and sea level rise in Palau to 2100. Natural Hazards and Earth System Sciences, 18, 669–686. https://doi.org/10.5194/ nhess-18-669-2018
- Hoozemans, F. M. J., Marchand, M., & Pennekamp, H. A. (1993). Sea Level rise: A global vulnerability assessment: Vulnerability assessments for population and coastal wetlands and rice production on a global scale, revised. Delft, The Hague, The Netherlands: Delft Hydraulics and Rijkswaterstaat.
- Huizinga, J., de Moel, H., & Szewczyk, W. (2017). *Global flood depth-damage functions: Methodology and the database with guidelines.* (No. EUR 28552). European Commission, Joint Research Centre.
- Hunter, J. (2012). A simple technique for estimating an allowance for uncertain sea-level rise. *Climatic Change*, 113, 239–252. https://doi.org/10.1007/s10584-011-0332-1
- Hunter, J. R., Church, J. A., White, N. J., & Zhang, X. (2013). Towards a global regionally varying allowance for sea-level rise. Ocean Engineering, Sea Level Rise and Impacts on Engineering Practice, 71, 17–27. https://doi.org/10.1016/j.oceaneng.2012.12.041
- Hunter, J. R., Woodworth, P. L., Wahl, T., & Nicholls, R. J. (2017). Using global tide gauge data to validate and improve the representation of extreme sea levels in flood impact studies. *Global and Planetary Change*, 156, 34–45. https://doi.org/10.1016/j.gloplacha.2017.06.007 Idier, D., Paris, F., Cozannet, G. L., Boulahya, F., & Dumas, F. (2017). Sea-level rise impacts on the tides of the European Shelf. *Continental*
- Shelf Research, 137, 56–71. https://doi.org/10.1016/j.csr.2017.0.007
- Ikeuchi, H., Hirabayashi, Y., Yamazaki, D., Muis, S., Ward, P. J., Winsemius, H. C., et al. (2017). Compound simulation of fluvial floods and storm surges in a global coupled river-coast flood model: Model development and its application to 2007 Cyclone Sidr in Bangladesh. *Journal of Advances in Modeling Earth Systems*, 9, 1847–1862. https://doi.org/10.1002/2017MS000943
- IPCC. (2014a). Climate change 2014: Impacts, adaptation, and vulnerability. Part A: Global and sectoral aspects. In Field, C. B., et al. (Eds.), Contribution of working group II to the fifth assessment report of the intergovernmental Panel on climate change. Cambridge, United Kingdom and NY, USA: Cambridge University Press.
- IPCC. (2014b). Climate change 2014: Impacts, adaptation, and vulnerability. Part B: Regional aspects. In Agard, J., et al. (Eds.), *Contribution of working group II to the fifth assessment report of the intergovernmental Panel on climate change*. Cambridge, United Kingdom and NY, USA: Cambridge University Press.
- Jones, B., & O'Neill, B. C. (2016). Spatially explicit global population scenarios consistent with the Shared Socioeconomic Pathways. Environmental Research Letters, 11, 084003. https://doi.org/10.1088/1748-9326/11/8/084003
- Jones, R. N., Patwardhan, A., Cohen, S. J., Dessai, S., Lammel, A., Lempert, R. J., et al. (2014). Foundations for decision making. In Field, C. B., et al. (Eds.), Climate change 2014: Impacts, adaptation, and vulnerability. Part A: Global and sectoral aspects. Contribution of working group II to the fifth assessment report of the intergovernmental Panel on climate change (p. XXX). Cambridge, United Kingdom and New York, NY, USA: Cambridge University Press.
- Jongman, B., Hochrainer-Stigler, S., Feyen, L., Aerts, J. C. J. H., Mechler, R., Botzen, W. J. W., et al. (2014). Increasing stress on disaster-risk finance due to large floods. *Nature Climate Change*, 4, 264–268. https://doi.org/10.1038/nclimate2124
- Jongman, B., Kreibich, H., Apel, H., Barredo, J. I., Bates, P. D., Feyen, L., et al. (2012). Comparative flood damage model assessment: Towards a European approach. Natural Hazards and Earth System Sciences, 12, 3733–3752. https://doi.org/10.5194/nhess-12-3733-2012
- Jongman, B., Winsemius, H. C., Aerts, J. C. J. H., de Perez, E. C., van Aalst, M. K., Kron, W., & Ward, P. J. (2015). Declining vulnerability to river floods and the global benefits of adaptation. *Proceedings of the National Academy of Sciences*, 112, E2271–E2280. https://doi. org/10.1073/pnas.1414439112
- KC, S., & Lutz, W. (2017). The human core of the shared socioeconomic pathways: Population scenarios by age, sex and level of education for all countries to 2100. *Global Environmental Change*, 42, 181–192. https://doi.org/10.1016/j.gloenvcha.2014.06.004
- Kellndorfer, J., Walker, W., Pierce, L., Dobson, C., Fites, J. A., Hunsaker, C., et al. (2004). Vegetation height estimation from Shuttle Radar Topography Mission and National Elevation Datasets. *Remote Sensing of Environment*, 93, 339–358. https://doi.org/10.1016/j.rse.2004.07.017
- Kirezci, E., Young, I. R., Ranasinghe, R., Muis, S., Nicholls, R. J., Lincke, D., & Hinkel, J. (2020). Projections of global-scale extreme sea levels and resulting episodic coastal flooding over the 21st Century. *Scientific Reports*, 10, 11629. https://doi.org/10.1038/s41598-020-67736-6

- Koivumäki, L., Alho, P., Lotsari, E., Käyhkö, J., Saari, A., & Hyyppä, H. (2010). Uncertainties in flood risk mapping: A case study on estimating building damages for a river flood in Finland: Uncertainties in flood risk mapping. *Journal of Flood Risk Management*, 3, 166–183. https://doi.org/10.1111/j.1753-318X.2010.01064.x
- Koks, E. E., Thissen, M., Alfieri, L., De Moel, H., Feyen, L., Jongman, B., & Aerts, J. C. J. H. (2019). The macroeconomic impacts of future river flooding in Europe. *Environmental Research Letters*, 14, 084042. https://doi.org/10.1088/1748-9326/ab3306
- Komen, G. J., Cavaleri, L., Donelan, M., Hasselmann, K., Hasselmann, S., & Janssen, P. A. E. M. (1996). In G. J. Komen, L. Cavaleri, M. Donelan, K. Hasselmann, S. Hasselmann, & P. A. E. M. Janssen (Eds.), *Dynamics and modelling of ocean waves*. (p. 554). Cambridge, UK: Cambridge University Press.
- Kopp, R. E., Horton, R. M., Little, C. M., Mitrovica, J. X., Oppenheimer, M., Rasmussen, D. J., et al. (2014). Probabilistic 21st and 22nd century sea-level projections at a global network of tide-gauge sites: KOPP ET AL. *Earth's Future*, 2, 383–406. https://doi. org/10.1002/2014EF000239
- Kreibich, H., Baldassarre, G. D., Vorogushyn, S., Aerts, J. C. J. H., Apel, H., Aronica, G. T., et al. (2017). Adaptation to flood risk: Results of international paired flood event studies. *Earth's Future*, 5, 953–965. https://doi.org/10.1002/2017EF000606
- Kriegler, E., O'Neill, B. C., Hallegatte, S., Kram, T., Lempert, R. J., Moss, R. H., & Wilbanks, T. (2012). The need for and use of socio-economic scenarios for climate change analysis: A new approach based on shared socio-economic pathways. *Global Environmental Change*, 22, 807–822. https://doi.org/10.1016/j.gloenvcha.2012.05.005
- Kulp, S. A., & Strauss, B. H. (2018). CoastalDEM: A global coastal digital elevation model improved from SRTM using a neural network. Remote Sensing of Environment, 206, 231–239. https://doi.org/10.1016/j.rse.2017.12.026
- Kulp, S. A., & Strauss, B. H. (2019). New elevation data triple estimates of global vulnerability to sea-level rise and coastal flooding. Nature Communications, 10, 1–12. https://doi.org/10.1038/s41467-019-12808-z
- Kunreuther, H., Heal, G., Allen, M., Edenhofer, O., Field, C. B., & Yohe, G. (2013). Risk management and climate change. Nature Climate Change, 3, 447–450. https://doi.org/10.1038/nclimate1740
- Lambert, E., Le Bars, D., Goelzer, H., & van de Wal, R. S. W. (2021). Correlations between sea-level components are driven by regional climate change. *Earth's Future*, 9. https://doi.org/10.1029/2020EF001825
- Lambert, E., Rohmer, J., Cozannet, G. L., & van de Wal, R. S. W. (2020). Adaptation time to magnified flood hazards underestimated when derived from tide gauge records. *Environmental Research Letters*, 15, 074015. https://doi.org/10.1088/1748-9326/ab8336
- Lazzaroni, S., & van Bergeijk, P. A. G. (2014). Natural disasters' impact, factors of resilience and development: A meta-analysis of the macroeconomic literature. *Ecological Economics*, 107, 333–346. https://doi.org/10.1016/j.ecolecon.2014.08.015
- Le Cozannet, G., Rohmer, J., Cazenave, A., Idier, D., van de Wal, R., de Winter, R., et al. (2015). Evaluating uncertainties of future marine flooding occurrence as sea-level rises. *Environmental Modelling & Software*, 73, 44–56. https://doi.org/10.1016/j.envsoft.2015.07.021
- Lichter, M., Vafeidis, A. T., Nicholls, R. J., & Kaiser, G. (2011). Exploring data-related uncertainties in analyses of land area and population in the "Low-Elevation Coastal Zone" (LECZ). Journal of Coastal Research, 27, 757–768. https://doi.org/10.2112/jcoastres-d-10-00072.1

Lin, N., & Emanuel, K. (2016). Grey swan tropical cyclones. Nature Climate Change, 6, 106–111. https://doi.org/10.1038/nclimate2777

Lincke, D., & Hinkel, J. (2018). Economically robust protection against 21st century sea-level rise. *Global Environmental Change*, 51, 67–73. https://doi.org/10.1016/j.gloenvcha.2018.05.003

- Little, C. M., Horton, R. M., Kopp, R. E., Oppenheimer, M., & Yip, S. (2015). Uncertainty in twenty-first-century CMIP5 sea level projections. Journal of Climate, 28, 838–852. https://doi.org/10.1175/JCLI-D-14-00453.1
- Luana, S., Hou, X., & Wang, Y. (2015). Assessing the accuracy of SRTM Dem and Aster Gdem datasets for the coastal zone of Shandong Province, Eastern China. Polish Maritime Research, 22, 15–20. https://doi.org/10.1515/pomr-2015-0026
- Luijendijk, A., Hagenaars, G., Ranasinghe, R., Baart, F., Donchyts, G., & Aarninkhof, S. (2018). The state of the world's beaches. Scientific Reports, 8. https://doi.org/10.1038/s41598-018-24630-6
- Mandel, A., Tiggeloven, T., Lincke, D., Koks, E., Ward, P., & Hinkel, J. (2020). Risks on global financial stability induced by climate change. (SSRN scholarly paper No. ID 3626936). Rochester, NY: Social Science Research Network. https://doi.org/10.2139/ssrn.3626936
- Marcos, M., Calafat, F. M., Berihuete, Á., & Dangendorf, S. (2015). Long-term variations in global sea level extremes. Journal of Geophysical Research: Oceans, 120, 8115–8134. https://doi.org/10.1002/2015JC011173
- Marcos, M., Rohmer, J., Vousdoukas, M. I., Mentaschi, L., Cozannet, G. L., & Amores, A. (2019). Increased extreme coastal water levels due to the combined action of storm surges and wind waves. *Geophysical Research Letters*, 46, 4356–4364. https://doi. org/10.1029/2019GL082599
- Marcos, M., Tsimplis, M. N., & Shaw, A. G. P. (2009). Sea level extremes in southern Europe. *Journal of Geophysical Research*, 114, C01007. https://doi.org/10.1029/2008JC004912
- McClean, F., Dawson, R., & Kilsby, C. (2020). Implications of using global digital elevation models for flood risk analysis in cities. Water Resources Research, 56, e2020WR028241. https://doi.org/10.1029/2020WR028241
- McEvoy, S., Haasnoot, M., & Biesbroek, R. (2021). How are European countries planning for sea level rise? Ocean & Coastal Management, 203, 105512. https://doi.org/10.1016/j.ocecoaman.2020.105512
- McGranahan, G., Balk, D., & Anderson, B. (2007). The rising tide: Assessing the risks of climate change and human settlements in low elevation coastal zones. *Environment and Urbanization*, 19, 17–37. https://doi.org/10.1177/0956247807076960

Melet, A., Almar, R., Hemer, M., Le Cozannet, G., Meyssignac, B., & Ruggiero, P. (2020). Contribution of wave setup to projected coastal sea level changes. *Journal of Geophysical Research: Oceans*, 125, e2020JC016078. https://doi.org/10.1029/2020JC016078

Melet, A., Meyssignac, B., Almar, R., & Le Cozannet, G. (2018). Under-estimated wave contribution to coastal sea-level rise. Nature Climate Change, 8, 234–239. https://doi.org/10.1038/s41558-018-0088-y

Menendez, M., Mendez, F. J., & Losada, I. J. (2009). Forecasting seasonal to interannual variability in extreme sea levels. ICES Journal of Marine Science, 66, 1490–1496. https://doi.org/10.1093/icesjms/fsp095

Menendez, M., & Woodworth, P. L. (2010). Changes in extreme high water levels based on a quasi-global tide-gauge dataset. Journal of Geophysical Research, 115, 1–15. https://doi.org/10.1029/2009JC005997

Mentaschi, L. (2018). The effect of changing spatial resolution in global dynamic wave models. *Earth and Space Science Open Archive*. https://doi.org/10.1002/essoar.10500014.1

- Mentaschi, L., Besio, G., Cassola, F., & Mazzino, A. (2015). Performance evaluation of Wavewatch III in the Mediterranean Sea. Ocean Modelling, 90, 82–94. https://doi.org/10.1016/j.ocemod.2015.04.003
- Mentaschi, L., Vousdoukas, M. I., Voukouvalas, E., Dosio, A., & Feyen, L. (2017). Global changes of extreme coastal wave energy fluxes triggered by intensified teleconnection patterns. *Geophysical Research Letters*, 44, 2416–2426. https://doi.org/10.1002/2016GL072488



Mentaschi, L., Vousdoukas, M., Montblanc, T. F., Kakoulaki, G., Voukouvalas, E., Besio, G., & Salamon, P. (2020). Assessment of global wave models on regular and unstructured grids using the Unresolved Obstacles Source Term. Ocean Dynamics, 70, 1475–1483. https:// doi.org/10.1007/s10236-020-01410-3

Mentaschi, L., Vousdoukas, M., Voukouvalas, E., Sartini, L., Feyen, L., Besio, G., & Alfieri, L. (2016). The transformed-stationary approach: A generic and simplified methodology for non-stationary extreme value analysis. *Hydrology and Earth System Sciences*, 20, 3527–3547. https://doi.org/10.5194/hess-20-3527-2016

- Merkens, J.-L., Lincke, D., Hinkel, J., Brown, S., & Vafeidis, A. T. (2018). Regionalisation of population growth projections in coastal exposure analysis. *Climatic Change*, 151, 413–426. https://doi.org/10.1007/s10584-018-2334-8
- Merkens, J.-L., Reimann, L., Hinkel, J., & Vafeidis, A. T. (2016). Gridded population projections for the coastal zone under the Shared Socioeconomic Pathways. Global and Planetary Change, 145, 57–66. https://doi.org/10.1016/j.gloplacha.2016.08.009
- Merrifield, M. A., Genz, A. S., Kontoes, C. P., & Marra, J. J. (2013). Annual maximum water levels from tide gauges: Contributing factors and geographic patterns. Journal of Geophysical Research: Oceans, 118, 2535–2546. https://doi.org/10.1002/jgrc.20173
- Meucci, A., Young, I. R., & Breivik, Ø. (2018). Wind and wave extremes from atmosphere and wave model ensembles. *Journal of Climate*, 31, 8819–8842. https://doi.org/10.1175/JCLI-D-18-0217.1
- Meucci, A., Young, I. R., Hemer, M., Kirezci, E., & Ranasinghe, R. (2020). Projected 21st century changes in extreme wind-wave events. Science Advances, 6, eaaz7295. https://doi.org/10.1126/sciadv.aaz7295
- Mondal, P., & Tatem, A. J. (2012). Uncertainties in measuring populations potentially impacted by sea level rise and coastal flooding. PLOS ONE, 7, e48191. https://doi.org/10.1371/journal.pone.0048191
- Morgan, M. G., Henrion, M., & Small, M. (1990). Uncertainty: A guide to dealing with uncertainty in quantitative risk and policy analysis. Cambridge University Press.
- Morim, J., Hemer, M., Cartwright, N., Strauss, D., & Andutta, F. (2018). On the concordance of 21st century wind-wave climate projections. Global and Planetary Change, 167, 160–171. https://doi.org/10.1016/j.gloplacha.2018.05.005
- Morim, J., Hemer, M., Wang, X. L., Cartwright, N., Trenham, C., Semedo, A., et al. (2019). Robustness and uncertainties in global multivariate wind-wave climate projections. *Nature Climate Change*, 9, 711–718. https://doi.org/10.1038/s41558-019-0542-5
- Muis, S., Apecechea, M. I., Dullaart, J., de Lima Rego, J., Madsen, K. S., Su, J., et al. (2020). A High-resolution global dataset of extreme sea levels, tides, and storm surges, including future projections. *Frontiers in Marine Science*, 7. https://doi.org/10.3389/fmars.2020.00263
- Muis, S., Verlaan, M., Nicholls, R. J., Brown, S., Hinkel, J., Lincke, D., et al. (2017). A comparison of two global datasets of extreme sea levels and resulting flood exposure. *Earth's Future*, 5, 379–392. https://doi.org/10.1002/2016EF000430
- Muis, S., Verlaan, M., Winsemius, H. C., Aerts, J. C. J. H., & Ward, P. J. (2016). A global reanalysis of storm surges and extreme sea levels. *Nature Communications*, 7, 11969. https://doi.org/10.1038/ncomms11969
- Neumann, B., Vafeidis, A. T., Zimmermann, J., & Nicholls, R. J. (2015). Future coastal population growth and exposure to sea-level rise and coastal flooding—A global assessment. PLOS ONE, 10, e0118571. https://doi.org/10.1371/journal.pone.0118571
- Nicholls, R. J., Brown, S., Goodwin, P., Wahl, T., Lowe, J., Solan, M., et al. (2018). Stabilization of global temperature at 1.5°C and 2.0°C: Implications for coastal areas. *Philosophical Transactions of the Royal Society A: Mathematical, Physical & Engineering Sciences, 376*, 20160448. https://doi.org/10.1098/rsta.2016.0448
- Nicholls, R. J., Hanson, S. E., Lowe, J. A., Warrick, R. A., Lu, X., & Long, A. J., (2014). Sea-level scenarios for evaluating coastal impacts. *WIREs Climate Change* 5, 129–150. https://doi.org/10.1002/wcc.253
- Nicholls, R. J., Hinkel, J., Lincke, D., & van der Pol, T. (2019). Global investment costs for coastal defense through the 21st century. (No. WPS8745). The World Bank.
- Nicholls, R. J., Lincke, D., Hinkel, J., Brown, S., Vafeidis, A. T., Meyssignac, B., et al. (2021). A global analysis of subsidence, relative sea-level change and coastal flood exposure. *Nature Climate Change*, 11, 338–342. https://doi.org/10.1038/s41558-021-00993-z
- O'Neill, B. C., Kriegler, E., Ebi, K. L., Kemp-Benedict, E., Riahi, K., Rothman, D. S., et al. (2017). The roads ahead: Narratives for shared socioeconomic pathways describing world futures in the 21st century. *Global Environmental Change*, 42, 169–180. https://doi.org/10.1016/j.gloenvcha.2015.01.004
- Oppenheimer, M., Glavovic, B., Hinkel, J., van de Wal, R., Magnan, A. K., Abd-Elgawad, A., et al. (2019). Sea level rise and implications for low lying islands, coasts and communities. In Special report on the ocean and cryosphere in a changing climate. Oxford University Press.
- Parodi, M. U., Giardino, A., van Dongeren, A., Pearson, S. G., Bricker, J. D., & Reniers, A. J. H. M. (2020). Uncertainties in coastal flood risk assessments in small island developing states. *Natural Hazards and Earth System Sciences*, 20, 2397–2414. https://doi.org/10.5194/ nhess-20-2397-2020
- Parrado, R., Bosello, F., Delpiazzo, E., Hinkel, J., Lincke, D., & Brown, S. (2020). Fiscal effects and the potential implications on economic growth of sea-level rise impacts and coastal zone protection. *Climatic Change*, 160, 283–302. https://doi.org/10.1007/s10584-020-02664-y
- Passeri, D. L., Hagen, S. C., Bilskie, M. V., & Medeiros, S. C. (2015). On the significance of incorporating shoreline changes for evaluating coastal hydrodynamics under sea level rise scenarios. *Natural Hazards*, 75, 1599–1617. https://doi.org/10.1007/s11069-014-1386-y
- Pattyn, F., Ritz, C., Hanna, E., Asay-Davis, X., DeConto, R., Durand, G., et al. (2018). The Greenland and Antarctic ice sheets under 1.5°C global warming. *Nature Climate Change*, 8, 1053–1061. https://doi.org/10.1038/s41558-018-0305-8
- Penning-Rowsell, E. (Ed.), (2013). Flood and coastal erosion risk management: A manual for economic appraisal. Milton Park, Abingdon, Oxon: Routledge.
- Penning-Rowsell, E., Priest, S., Parker, D., Morris, J., Tunstall, S., Viavattene, C., et al. (2014). Flood and coastal erosion risk management: A manual for economic Appraisal. Routledge.
- Pesaresi, M., Huadong, G., Blaes, X., Ehrlich, D., Ferri, S., Gueguen, L., et al. (2013). A global human settlement layer from optical HR/ VHR RS data: Concept and first results. *IEEE Journal of Selected Topics in Applied Earth Observations and Remote Sensing*, *6*, 2102–2131. https://doi.org/10.1109/JSTARS.2013.2271445
- Petter, G., Mairota, P., Albrich, K., Bebi, P., Brůna, J., Bugmann, H., et al. (2020). How robust are future projections of forest landscape dynamics? Insights from a systematic comparison of four forest landscape models. *Environmental Modelling & Software*, 134, 104844. https://doi.org/10.1016/j.envsoft.2020.104844
- Pickering, M. D., Horsburgh, K. J., Blundell, J. R., Hirschi, J. J.-M., Nicholls, R. J., Verlaan, M., & Wells, N. C. (2017). The impact of future sea-level rise on the global tides. *Continental Shelf Research*, 142, 50–68. https://doi.org/10.1016/j.csr.2017.02.004
- Poulter, B., & Halpin, P. N. (2008). Raster modelling of coastal flooding from sea-level rise. International Journal of Geographical Information Science, 22, 167–182. https://doi.org/10.1080/13658810701371858
- Prahl, B. F., Boettle, M., Costa, L., Kropp, J. P., & Rybski, D. (2018). Damage and protection cost curves for coastal floods within the 600 largest European cities. *Scientific Data*, 5, 180034. https://doi.org/10.1038/sdata.2018.34

- Ramirez, J. A., Lichter, M., Coulthard, T. J., & Skinner, C. (2016). Hyper-resolution mapping of regional storm surge and tide flooding: Comparison of static and dynamic models. *Natural Hazards*, 82, 571–590. https://doi.org/10.1007/s11069-016-2198-z
- Rashid, M. M., & Wahl, T. (2020). Predictability of extreme sea level variations along the U.S. coastline. Journal of Geophysical Research: Oceans, 125, e2020JC016295. https://doi.org/10.1029/2020JC016295
- Rashid, M. M., Wahl, T., Chambers, D. P., Calafat, F. M., & Sweet, W. V. (2019). An extreme sea level indicator for the contiguous United States coastline. *Scientific Data*, 6, 326. https://doi.org/10.1038/s41597-019-0333-x
- Rasmussen, D. J., Kopp, R. E., Shwom, R., & Oppenheimer, M. (2021). The political complexity of coastal flood risk reduction: Lessons for climate adaptation public works in the U.S. *Earth's Future*, 9, e2020EF001575. https://doi.org/10.1029/2020EF001575
- Rodriguez, E., Morris, C. S., Belz, J. E., Chapin, E. C., Martin, J. M., Daffer, W., & Hensley, S. (2005). An assessment of the SRTM topographic products (No. JPL D-31639). Pasadena: Jet Propulsion Laboratory.
- Roland, A., Zhang, Y. J., Wang, H. V., Meng, Y., Teng, Y.-C., Maderich, V., et al. (2012). A fully coupled 3D wave-current interaction model on unstructured grids. *Journal of Geophysical Research*, 117, C00J33. https://doi.org/10.1029/2012jc007952
- Rueda, A., Vitousek, S., Camus, P., Tomás, A., Espejo, A., Losada, I. J., et al. (2017). A global classification of coastal flood hazard climates associated with large-scale oceanographic forcing. *Scientific Reports*, 7. https://doi.org/10.1038/s41598-017-05090-w
- Saint-Geours, N., Grelot, F., Bailly, J.-S., & Lavergne, C. (2015). Ranking sources of uncertainty in flood damage modelling: A case study on the cost-benefit analysis of a flood mitigation project in the Orb Delta, France: Ranking sources of uncertainty in flood damage modelling. Journal of Flood Risk Management, 8, 161–176. https://doi.org/10.1111/jfr3.12068
- Saltelli, A., Ratto, M., Andres, T., Campolongo, F., Cariboni, J., Gatelli, D., et al. (2008). Global sensitivity analysis: The primer. John Wiley & Sons.
- Saulter, A., Bunney, C., Li, J.-G., & Palmer, T. (2017). Process and resolution impacts on UK coastal wave predictions from operational global-regional wave models. In Presented at the 15th international workshop on wave hindcasting and forecasting & 6th coastal hazard symposium (Vol. 10–15, p. 26). Liverpool, UK.
- Scawthorn, C., Flores, P., Blais, N., Seligson, H., Tate, E., Chang, S., et al. (2006). HAZUS-MH flood loss estimation methodology. II. Damage and loss assessment. Natural Hazards Review, 7, 72–81. https://doi.org/10.1061/(asce)1527-6988(2006)7:2(72)
- Schinko, T., Drouet, L., Vrontisi, Z., Hof, A., Hinkel, J., Mochizuki, J., et al. (2020). Economy-wide effects of coastal flooding due to sea level rise: A multi-model simultaneous treatment of mitigation, adaptation, and residual impacts. *Environmental Research Communication*, 2, 015002. https://doi.org/10.1088/2515-7620/ab6368
- Schmitt, F. G., Crapoulet, A., Hequette, A., & Huang, Y. (2018). Nonlinear dynamics of the sea level time series in the eastern English Channel. Natural Hazards, 91, 267–285. https://doi.org/10.1007/s11069-017-3125-7
- Schumann, G. J.-P., & Bates, P. D. (2018). The need for a high-accuracy, open-access global DEM. Frontiers of Earth Science, 6. https://doi. org/10.3389/feart.2018.00225
- Scussolini, P., Aerts, J. C. J. H., Jongman, B., Bouwer, L. M., Winsemius, H. C., de Moel, H., & Ward, P. J. (2016). FLOPROS: An evolving global database of flood protection standards. *Natural Hazards and Earth System Sciences*, 16, 1049–1061. https://doi.org/10.5194/ nhess-16-1049-2016
- Scussolini, P., Tran, T. V. T., Koks, E., Diaz-Loaiza, A., Ho, P. L., & Lasage, R. (2017). Adaptation to sea level rise: A multidisciplinary analysis for Ho Chi Minh City, Vietnam. Water Resources Research, 53, 10841–10857. https://doi.org/10.1002/2017WR021344
- Seenath, A., Wilson, M., & Miller, K. (2016). Hydrodynamic versus GIS modelling for coastal flood vulnerability assessment: Which is better for guiding coastal management? Ocean & Coastal Management, 120, 99–109. https://doi.org/10.1016/j.ocecoaman.2015.11.019
- Shirzaei, M., Freymueller, J., Törnqvist, T. E., Galloway, D. L., Dura, T., & Minderhoud, P. S. J. (2020). Measuring, modelling and projecting coastal land subsidence. *Nature Reviews Earth & Environment*, *2*, 40–58. https://doi.org/10.1038/s43017-020-00115-x
- Simpson, M., James, R., Hall, J. W., Borgomeo, E., Ives, M. C., Almeida, S., et al. (2016). Decision analysis for management of natural hazards. Annual Review of Environment and Resources, 41, 489–516. https://doi.org/10.1146/annurev-environ-110615-090011
- Slangen, A. B. A., Carson, M., Katsman, C. A., van de Wal, R. S. W., Köhl, A., Vermeersen, L. L. A., & Stammer, D. (2014). Projecting twenty-first century regional sea-level changes. *Climatic Change*, 124, 317–332. https://doi.org/10.1007/s10584-014-1080-9
- Stockdon, H. F., Holman, R. A., Howd, P. A., & Sallenger, A. H. (2006). Empirical parameterization of setup, swash, and runup. Coastal Engineering, 53, 573–588. https://doi.org/10.1016/j.coastaleng.2005.12.005
- Stopa, J. E. (2018). Wind forcing calibration and wave hindcast comparison using multiple reanalysis and merged satellite wind datasets. *Ocean Modelling*, 127, 55–69. https://doi.org/10.1016/j.ocemod.2018.04.008
- Swain, J., Umesh, P. A., Balchand, A. N., & Kumar, B. P. (2017). Wave hindcasting using WAM and WAVEWATCH III: A comparison study utilizing Oceansat-2 (OSCAT) winds. Journal of Oceanography and Marine Research, 5. https://doi.org/10.4172/2572-3103.1000166
- Syvitski, J. P. M., Kettner, A. J., Overeem, I., Hutton, E. W. H., Hannon, M. T., Brakenridge, G. R., et al. (2009). Sinking deltas due to human activities. *Nature Geoscience*, *2*, 681–686. https://doi.org/10.1038/ngeo629
- Tadesse, M., Wahl, T., & Cid, A. (2020). Data-driven modeling of global storm surges. Frontiers in Marine Science, 7. https://doi.org/10.3389/fmars.2020.00260
- Tamura, M., Kumano, N., Yotsukuri, M., & Yokoki, H. (2019). Global assessment of the effectiveness of adaptation in coastal areas based on RCP/SSP scenarios. *Climatic Change*, 152, 363–377. https://doi.org/10.1007/s10584-018-2356-2
- Tatem, A. J. (2017). WorldPop, open data for spatial demography. Scientific Data, 4, 170004. https://doi.org/10.1038/sdata.2017.4
- Tedstone, A. J., Bamber, J. L., Cook, J. M., Williamson, C. J., Fettweis, X., Hodson, A. J., & Tranter, M. (2017). Dark ice dynamics of the south-west Greenland Ice Sheet. *The Cryosphere*, 11, 2491–2506. https://doi.org/10.5194/tc-11-2491-2017
- Tiggeloven, T., de Moel, H., Winsemius, H. C., Eilander, D., Erkens, G., Gebremedhin, E., et al. (2020). Global-scale benefit-cost analysis of coastal flood adaptation to different flood risk drivers using structural measures. *Natural Hazards and Earth System Sciences*, 20, 1025–1044. https://doi.org/10.5194/nhess-20-1025-2020
- Timmermans, B., Stone, D., Wehner, M., & Krishnan, H. (2017). Impact of tropical cyclones on modeled extreme wind-wave climate. Geophysical Research Letters, 44, 1393–1401. https://doi.org/10.1002/2016GL071681
- Tol, R. S. J. (2007). The double trade-off between adaptation and mitigation for sea level rise: An application of FUND. *Mitigation and Adaptation Strategies for Global Change*, *12*, 741–753. https://doi.org/10.1007/s11027-007-9097-2
- Tolman, H. L. (2003). Treatment of unresolved islands and ice in wind wave models. *Ocean Modelling*, 5, 219–231. https://doi.org/10.1016/S1463-5003(02)00040-9
- Tolman, H. L. (2009). User manual and system documentation of WAVEWATCH III TM version 3.14 220.
- UNDRR. (2019). Global assessment report on disaster risk Reduction. UN Office for Disaster Risk Reduction.
- UNEP. (2016). The adaptation finance gap report 2016. Nairobi, Kenya: United Nations Environment Programme (UNEP). UNFCCC. (1992). United Nations framework convention on climate change.



- van de Wal, R. S. W., Zhang, X., Minobe, S., Jevrejeva, S., Riva, R. E. M., Little, C., et al. (2019). Uncertainties in long-term twenty-first century process-based coastal sea-level projections. *Surveys in Geophysics*, 40, 1655–1671. https://doi.org/10.1007/s10712-019-09575-3
  Vernimmen, R., Hooijer, A., & Pronk, M. (2020). New ICESat-2 satellite LiDAR data allow first global lowland DTM Suitable for accurate
- coastal flood risk assessment. *Remote Sensing*, *12*, 2827. https://doi.org/10.3390/rs12172827 Vitousek, S., Barnard, P. L., Fletcher, C. H., Frazer, N., Erikson, L., & Storlazzi, C. D. (2017). Doubling of coastal flooding frequency within decades due to sea-level rise. *Scientific Reports*, *7*. https://doi.org/10.1038/s41598-017-01362-7
- Vollset, S. E., Goren, E., Yuan, C.-W., Cao, J., Smith, A. E., Hsiao, T., et al. (2020). Fertility, mortality, migration, and population scenarios for 195 countries and territories from 2017 to 2100: A forecasting analysis for the Global Burden of Disease Study. *The Lancet, 396*, 1285–1306. https://doi.org/10.1016/S0140-6736(20)30677-2
- Vousdoukas, M. I., Bouziotas, D., Giardino, A., Bouwer, L. M., Mentaschi, L., Voukouvalas, E., & Feyen, L. (2018a). Understanding epistemic uncertainty in large-scale coastal flood risk assessment for present and future climates. *Natural Hazards and Earth System Sciences*, 18, 2127–2142. https://doi.org/10.5194/nhess-18-2127-2018
- Vousdoukas, M. I., Mentaschi, L., Hinkel, J., Ward, P. J., Mongelli, I., Ciscar, J.-C., & Feyen, L. (2020a). Economic motivation for raising coastal flood defenses in Europe. *Nature Communications*, 11, 1–11. https://doi.org/10.1038/s41467-020-15665-3
- Vousdoukas, M. I., Mentaschi, L., Voukouvalas, E., Bianchi, A., Dottori, F., & Feyen, L. (2018b). Climatic and socioeconomic controls of future coastal flood risk in Europe. Nature Climate Change, 8, 776–780. https://doi.org/10.1038/s41558-018-0260-4
- Vousdoukas, M. I., Mentaschi, L., Voukouvalas, E., Verlaan, M., & Feyen, L. (2017). Extreme sea levels on the rise along Europe's coasts. *Earth's Future*, 5, 304–323. https://doi.org/10.1002/2016EF000505
- Vousdoukas, M. I., Mentaschi, L., Voukouvalas, E., Verlaan, M., Jevrejeva, S., Jackson, L. P., & Feyen, L. (2018c). Global probabilistic projections of extreme sea levels show intensification of coastal flood hazard. *Nature Communications*, 9, 2360. https://doi.org/10.1038/ s41467-018-04692-w
- Vousdoukas, M. I., Ranasinghe, R., Mentaschi, L., Plomaritis, T. A., Athanasiou, P., Luijendijk, A., & Feyen, L. (2020b). Sandy coastlines under threat of erosion. *Nature Climate Change*, 10, 260–263. https://doi.org/10.1038/s41558-020-0697-0
- Vousdoukas, M. I., Voukouvalas, E., Annunziato, A., Giardino, A., & Feyen, L. (2016a). Projections of extreme storm surge levels along Europe. Climate Dynamics, 47, 3171–3190. https://doi.org/10.1007/s00382-016-3019-5
- Vousdoukas, M. I., Voukouvalas, E., Mentaschi, L., Dottori, F., Giardino, A., Bouziotas, D., et al. (2016b). Developments in large-scale coastal flood hazard mapping. Natural Hazards and Earth System Sciences Discussions, 1–24. https://doi.org/10.5194/nhess-2016-124
- Wadey, M., Brown, S., Nicholls, R. J., & Haigh, I. (2017). Coastal flooding in the Maldives: An assessment of historic events and their implications. *Natural Hazards*, 89, 131–159. https://doi.org/10.1007/s11069-017-2957-5
- Wadey, M. P., Nicholls, R. J., & Hutton, C. (2012). Coastal flooding in the solent: An integrated analysis of defences and inundation. *Water*, 4, 430–459. https://doi.org/10.3390/w4020430
- Wagenaar, D. J., de Bruijn, K. M., Bouwer, L. M., & de Moel, H. (2016). Uncertainty in flood damage estimates and its potential effect on investment decisions. Natural Hazards and Earth System Sciences, 16, 1–14. https://doi.org/10.5194/nhess-16-1-2016
- Wahl, T., Haigh, I. D., Nicholls, R. J., Arns, A., Dangendorf, S., Hinkel, J., & Slangen, A. B. A. (2017). Understanding extreme sea levels for broad-scale coastal impact and adaptation analysis. *Nature Communications*, 8, 16075. https://doi.org/10.1038/ncomms16075
- Wandres, M., Pattiaratchi, C., & Hemer, M. A. (2017). Projected changes of the southwest Australian wave climate under two atmospheric greenhouse gas concentration pathways. Ocean Modelling, 117, 70–87. https://doi.org/10.1016/j.ocemod.2017.08.002
- Wang, X. L., Feng, Y., & Swail, V. R. (2014). Changes in global ocean wave heights as projected using multimodel CMIP5 simulations. Geophysical Research Letters, 41, 1026–1034. https://doi.org/10.1002/2013GL058650
- Wang, X. L., Feng, Y., & Swail, V. R. (2015). Climate change signal and uncertainty in CMIP5-based projections of global ocean surface wave heights. Journal of Geophysical Research: Oceans, 120, 3859–3871. https://doi.org/10.1002/2015JC010699
- Winter, B., Schneeberger, K., Huttenlau, M., & Stötter, J. (2018). Sources of uncertainty in a probabilistic flood risk model. *Natural Hazards*, 91, 431–446. https://doi.org/10.1007/s11069-017-3135-5
- Wolff, C., Vafeidis, A. T., Lincke, D., Marasmi, C., & Hinkel, J. (2016). Effects of scale and input data on assessing the future impacts of coastal flooding: An application of DIVA for the Emilia-Romagna Coast. Frontiers in Marine Science, 3. https://doi.org/10.3389/ fmars.2016.00041
- Wong, P. P., Losada, I. J., Gattuso, J.-P., Hinkel, J., Khattabi, A., McInnes, K. L., et al. (2014). Coastal systems and low-lying areas. In Field, C. B., et al. (Eds.), Climate change 2014: Impacts, adaptation, and vulnerability. Part A: Global and sectoral aspects. Contribution of working group II to the fifth assessment report of the intergovernmental Panel of climate change (pp. 361–409). Cambridge, United Kingdom and NY, USA: Cambridge University Press.
- Woodruff, J. D., Irish, J. L., & Camargo, S. J. (2013). Coastal flooding by tropical cyclones and sea-level rise. Nature, 504, 44–52. https://doi. org/10.1038/nature12855
- Woodworth, P. L., Hunter, J. R., Marcos, M., Caldwell, P., Menéndez, M., & Haigh, I. (2016). Towards a global higher-frequency sea level dataset. *Geoscience Data Journal*, 3, 50–59. https://doi.org/10.1002/gdj3.42
- Yamazaki, D., Ikeshima, D., Neal, J. C., O'Loughlin, F., Sampson, C. C., Kanae, S., & Bates, P. D. (2017). MERIT DEM: A new high-accuracy global digital elevation model and its merit to global hydrodynamic modeling. In AGU fall meeting abstracts.
- Yohe, G., & Tol, R. S. J. (2002). Indicators for social and economic coping capacity—Moving toward a working definition of adaptive capacity. *Global Environmental Change*, *12*, 25–40. https://doi.org/10.1016/S0959-3780(01)00026-7
- Zaherpour, J., Gosling, S. N., Mount, N., Schmied, H. M., Veldkamp, T. I. E., Dankers, R., et al. (2018). Worldwide evaluation of mean and extreme runoff from six global-scale hydrological models that account for human impacts. *Environmental Research Letters*, *13*, 065015. https://doi.org/10.1088/1748-9326/aac547
- Zijl, F., Verlaan, M., & Gerritsen, H. (2013). Improved water-level forecasting for the Northwest European Shelf and North Sea through direct modelling of tide, surge and non-linear interaction. Ocean Dynamics, 63, 823–847. https://doi.org/10.1007/s10236-013-0624-2